

Article



Analysis of Sources of Large Positioning Errors in Deterministic Fingerprinting

Joaquín Torres-Sospedra^{1,*,†} and Adriano Moreira^{2,*,†}

- ¹ Institute of New Imaging Technologies, Universitat Jaume I, 12071 Castellón de la Plana, Spain
- ² Algoritmi Research Centre, University of Minho, 4800-058 Guimarães, Portugal
- * Correspondence: jtorres@uji.es (J.T.-S.); adriano.moreira@algoritmi.uminho.pt (A.M.)
- + These authors contributed equally to this work.

Received: 31 October 2017; Accepted: 23 November 2017; Published: 27 November 2017

Abstract: Wi-Fi fingerprinting is widely used for indoor positioning and indoor navigation due to the ubiquity of wireless networks, high proliferation of Wi-Fi-enabled mobile devices, and its reasonable positioning accuracy. The assumption is that the position can be estimated based on the received signal strength intensity from multiple wireless access points at a given point. The positioning accuracy, within a few meters, enables the use of Wi-Fi fingerprinting in many different applications. However, it has been detected that the positioning error might be very large in a few cases, which might prevent its use in applications with high accuracy positioning requirements. Hybrid methods are the new trend in indoor positioning since they benefit from multiple diverse technologies (Wi-Fi, Bluetooth, and Inertial Sensors, among many others) and, therefore, they can provide a more robust positioning accuracy. In order to have an optimal combination of technologies, it is crucial to identify when large errors occur and prevent the use of extremely bad positioning estimations in hybrid algorithms. This paper investigates why large positioning errors occur in Wi-Fi fingerprinting and how to detect them by using the received signal strength intensities.

Keywords: indoor positioning; Wi-Fi fingerprinting; simulation; positioning errors

1. Introduction

It is well known that tracking, localization and navigation are interesting topics in academia and industry. They all have a common feature: the position of someone or something needs to be known. Positioning is valuable for end-users, tracking and advertising companies, government, and service providers, among many other involved actors. People can record their sporting activities and share them on social media [1]; companies can passively track people inside commercial areas [2,3] and analyse the customer patterns [4]; cars and robots can autonomously navigate and be tracked [5–7]. In recent years, location-based services (LBS) and real-time location systems (RTLS) have grown. The daily use of many smartphone applications, which require positioning, has been one reason for their growth. In fact, the LBS and RTLS market is expected to be USD 77.84 Billion by 2021 to support local search, recommender systems, e-commerce and business intelligence, among others.

This paper is focused on positioning [8], which is a term commonly associated to Global Navigation Satellite Systems (GNSS) such as GPS, Gallileo, GLONASS or BEIDU. However, there are some scenarios where GNSS cannot operate adequately and other solutions are required to support positioning. Indoor environments are of special interest since many studies show that people spend most of their time indoors [9–12], but GNSS signals do not penetrate structural elements and obstacles with ease. The GNSS solutions that work outdoors might not properly work indoors because radio signals can be easily blocked, attenuated or reflected [13].

In contrast to outdoor scenarios, there are many solutions and technologies to provide positioning indoors [14,15] and the deployment of a particular indoor positioning system depends on the operating

area [16]. Crucial decisions about the base technologies and techniques are taken when systems are being designed and a trade-off among accuracy, costs, ubiquity, availability, scalability and other features is carefully considered [16]. Although accuracy is the key objective to be optimized, the priorities depend on the scenario. Autonomous robot navigation might require high precision approaches, whereas the positioning error may be less important in pedestrian-based applications. Thus, the requirements, the base technology and the deployment are different in both scenarios.

Wi-Fi fingerprinting is a well known indoor positioning technique, which consists of two stages: calibration (off-line phase) and operation (on-line phase) [17–19]. The former stage is related to the site survey where the received signal strength intensities (RSSI values) from multiple wireless access points (WAPs or simply APs) are recorded at predefined well-known positions. Each record is called a Wi-Fi fingerprint and it can be considered a snapshot of the Wi-Fi signal strength from the multiple APs received at a given point. A calibration database, or radio map, is generated with all the fingerprints collected at this first stage across the operational area. The latter stage, operation, is related to the positioning itself. A fingerprint is collected at an unknown point and a matching algorithm is applied to estimate its position. This matching algorithm estimates the position by using the known positions from the most similar fingerprints stored in the calibration database. The terms Wi-Fi, WiFi, 802.11 and WLAN are commonly used as synonyms in the literature.

Wi-Fi fingerprinting assumes that all fingerprints, and therefore the corresponding RSSI values, are similar at a given point over time. Typical mean positioning errors are in the range of a few meters [20], which is enough for pedestrian indoor positioning and navigation in most of the cases. The observed mean error highly depends on the matching algorithm used and the scenario. Heterogeneity of hardware, density and distribution of calibration and operational points, density and distribution of APs, and the materials present in the environment, among many other factors, affect the mean positioning error. The consequence is that the same indoor positioning algorithm might report different accuracies in different areas [21]. However, unexpected and unacceptably large positioning errors are often observed in working systems deployed in real environments, even when the mean positioning error is low, as depicted in the examples shown in Figure 1. Figure 1 shows the Cumulative Distribution Function (CDF) for 1 kNN-based IPS using a database collected at the DSI department (University of Minho, Guimarães, Portugal), a Library area (Universitat Jaume I, Castellón, Spain) and the UMinho systems that participated in the 2015 and 2016 IPIN competitions [21,22]. In all cases, the percentage of operational fingerprints that are reporting a positioning error higher than twice the average error is significant (higher than 10% approx).

Although the vast majority of positioning errors are around the mean value, there are always a few cases whose positioning error is much larger, e.g., the UMinho IPS provided positioning errors higher than 20 m in the 2015 and 2016 IPIN competition (see Figure 1). While these large errors might be acceptable in some application areas, such as pedestrian navigation in shopping malls, other applications, such as robot navigation or tracking of vehicles for autonomous driving [23], are more demanding and do not tolerate these large errors.

In an attempt to leverage the attractive characteristics of fingerprinting while trying to circumvent its limited accuracy, some researchers have proposed hybrid positioning methods where fingerprinting is combined with other techniques. Examples of this approach are the combination of fingerprinting with pedestrian dead reckoning [24]; map matching and the physical structure of the buildings [25]; and Bluetooth [26]. Across these works, several combining techniques are also used, with the Kalman and Particle filters being the most frequent ones [27]. In this context, knowledge about the statistics of the errors enables the optimal combination of the position estimates provided by the merged solutions.



Figure 1. Cumulative distribution of the positioning error (CDF) for four different cases. Simple Wi-Fi fingerprinting system based on kNN at the DSI department (University of Minho, Portugal); Simple Wi-Fi fingerprinting system based on kNN at a small area of the university library building (Universiat Jaume I, Spain); UMinho system at the 2015 IPIN competition; UMinho system at the 2016 IPIN competition. The errors and penalties in floor detection are not considered in any of the results shown. Dashed vertical lines indicate the average error, whereas dotted vertical lines indicate twice the average error.

While the general trends of the statistics of errors in Wi-Fi fingerprinting are well known (e.g., see Figure 1), estimating the error associated with each individual position estimate is still a challenge and an open research topic. Of particular interest is the detection of cases where the errors are very large. So far, this research challenge has only been addressed by a few researchers, among them the works done by Depster et al. [28], Lemelson et al. [29], Farshad et al. [30], Zhi-An Deng [31] and Berkvens et al. [32].

Montgomery established that the nuisance factors in an experimental setup can be controlled, uncontrolled or, even, unknown [33]. This paper analyses how many controlled and uncontrolled factors have a direct impact on Wi-Fi fingerprinting accuracy: grid size; density and location of APs; density of fingerprints and positioning algorithm and its parameters. The "unknown" factor is also covered by assuming that the RSSI values (and, therefore, the noise present in the RSSI readings in a given position) follow a Gaussian distribution. Although this assumption is not always true, it significantly simplifies the computations with little performance difference according to [34]. In particular, the main objective of this paper is to explore why large positioning errors occur and how they can be detected by using only the information of the fingerprints, i.e., the RSSI values.

The rest of this paper is organized as follows. Section 2 reviews the related work on exploring and reducing the positioning error. Section 3 describes fingerprinting. Section 4 introduces the comprehensive study of the sources of (large) positioning errors in deterministic fingerprinting. Section 5 shows the relation observed between large errors and fingerprints and the results on real scenarios. Section 6 presents the conclusions drawn from this work.

2. Related Work

This section reviews the existing works that address the reduction of positioning errors in Wi-Fi fingerprinting.

Kaemarungsi and Krishnamurthy [35] developed a framework for analyzing a simple deterministic indoor positioning system based on the Euclidean distance to compare and match the fingerprints. The authors claimed that the accuracy of a positioning system could be determined, in terms of the probability of estimating the correct position, from the number of access points, grid spacing and path loss exponent. However, the analysis performed did not consider the accuracy and

precision in the position estimate. Later, in 2012, they presented a statistical data analysis of the RSSI values in [36]. One of the main findings was that positioning errors were proportional to the increase of the standard deviation of RSSI values, which depended on the WLAN card and scenario.

Youssef and Agrawala [37] presented an analytical method in order to obtain the optimal strategy for selecting the user position. An analytical analysis was also performed to study the impact of averaging multiple fingerprints on accuracy. This work showed that the accuracy of their indoor positioning system was consistent under different user profiles when the number of visible access points was reasonable at each possible position. One of the main objectives of the work done by Youssef and Agrawala was to find the function that minimized the probability of positioning error at the operational stage.

Krishnakumar and Krishnan [38] linked the minimum uncertainty with a lower hit on the median error and they also made some important observations about the dependence of the estimation uncertainty and various factors: signal variance, number of APs, distance between the APs and the signal propagation constant.

Wallbaum [39] used some analytical models and the empirical multi-wall radio propagation model [40] to investigate the influence of the following main parameters on indoor positioning: RSS deviation, number of visible access points, grid granularity and AP geometry. This work presented a comparative study of analytical and experimental results, which showed that both captured the qualitative behaviour of the real system. Wallbaum concluded that it should be investigated whether the models actually represent the upper and lower bounds of the positioning error.

Dempster et al. [28] introduced an analysis of RSSI variance, where it was shown that the user's orientation at a reference point influences the RSSI value captured by a device. Averaging fingerprints just for position (i.e., the measurements taken at the same reference point without considering the user's orientation) had higher variance than averaging the fingerprints per position and orientation, so averaging by position and orientation was more discriminative. Moreover, this paper concluded that the relationship between the real and RSSI distances of two fingerprints was poor. Comparing fingerprints with a not totally well-suited distance metric was suggested to be one of the reasons for positioning error. Finally, the authors highlighted that the physical underpinning might be easily ignored in dBm-based fingerprinting, being the RSSI values wrongly treated as unitless numbers.

Lemelson et al. [29] proposed four algorithms for estimating the error with just the information from fingerprints and improved the overall accuracy of the IPS proposed by Haeberlen et al. [41].

- 1. Applying efficient fingerprint clustering and using the cluster area information as a measure of the error (the real position is in the cluster in 90% of cases);
- 2. Using Leaving One (Out) fingerprint to compute an averaged map with the positioning error at well-known positions;
- 3. Using the average distance among all the *k* best candidates at the operational stage (the *k* references fingerprints that are most similar to the operational one) as the error estimation;
- 4. Using the variance of RSSI measurements at the operational stage to estimate the position error.

The results showed that a combination of the four algorithms significantly improved the accuracy of the IPS in two scenarios. However, this work did not introduce any study or analytical model in order to understand the origins of the (large) positioning errors as had been done in previous works [35,37–39].

Farshan et al. [30] took a microscopic look at Wi-Fi fingerprinting using smartphones. They firstly introduced seven definitions to improve the matching process at the operational stage using a deterministic approach (e.g., Radar [17,42]) and a probabilistic-based approach (e.g., Horus [37,43]). Three different distance metrics were considered in the former approach and two well-known distributions for RSSI modelling were used in the latter. The results showed that the combination of fingerprint definition and estimation algorithm that yields the best positioning accuracy highly depends on the environment and even on the floor for a given large environment. Moreover, they

analysed the impact of frequency band on fingerprinting. The RSSI is more stable for the 5 GHz band, which enables the IPS to provide better positioning accuracy. The 5 GHz band is less affected by temporal signal variations due to people movement because the probes are sent at a higher bit-rate than the 2.4 GHz band probes. Also, the low variability is also due to the low co-channel interference in the 5 GHz band. Finally, the presence of Virtual APs (physical devices emitting multiple Wi-Fi networks) also reduces the positioning error because the AP density increases, which seems to have a

positive correlation with accuracy, and the temporal variability inherent to wireless signal propagation is better captured. All results were provided as CDFs but they were truncated to errors of up to 10 m (large errors were omitted).

Zhi-An Deng et al. [31] proposed an indoor positioning system that exploits the data from Wi-Fi, smartphone sensors, landmarks and user motion status. One crucial step in their system was to detect outliers in RSS values, i.e., operational fingerprints that might have attached a large positioning error. An extended Kalman filter fusion approach might be negatively affected by an estimation with a large positioning error which can propagate to future position estimations, degrading positioning accuracy. Zhi-an Deng et al. proposed using the averaged weight of the calibration points in the trusted area as an indication of RSS reliability.

Berkvens et al. [32] studied the uncertainty of a Wi-Fi fingerprinting positioning system by relating the positioning error with the conditional entropy in the position posterior probability distribution. The uncertainty is commonly calculated as the positioning error using samples with well-known ground truth. Berkvens et al. proposed using the conditional entropy of a posterior probability distribution as a complementary measure of uncertainty, which had the advantages of being dynamic and not requiring ground truth. Based on this, they proposed a sensor model where the conditional entropy is an indicator of the upper bound of the positioning error for a particular operational estimate, i.e., a low conditional entropy value means that the positioning error is low, whereas a high value means that the positioning error might be high.

In general, most of the papers that have dealt with knowing the origins of the positioning error have the objective of reducing the average positioning error or showing a few features that originate from the positioning error. This paper is focused on investigating where the (very) large errors occur; as far as we know, there is no previous paper with this main objective.

3. Wi-Fi Fingerprinting

Wi-Fi fingerprinting is a well-known indoor positioning solution which relies on a fundamental assumption: the Wi-Fi signals measured in the environment have a unique signature—the Wi-Fi fingerprint—at a given position.

Wi-Fi fingerprinting has two differentiated phases: calibration (or training) and operation. In the calibration phase, which occurs before the operational phase, the environment is surveyed in order to create a radio map (also known as calibration or reference dataset). This process is empirically done by collecting Wi-Fi fingerprints at different reference points whose positions are well known. Regarding this stage, there is no clear best strategy and slight differences and assumptions can be found in the already proposed systems. Later, during the operational phase, a new fingerprint is collected with the signal strength of all detected APs at an unknown position. This fingerprint is compared with the fingerprints in the radio map and the position is computed using the positions of the most similar reference fingerprints.

Wi-Fi technology was initially designed to support communications, not for positioning. The propagation of radio signals in an indoor environment is not easy to predict due to the presence of people [35,44] and existing obstacles, which create reflections, refractions and multipath interference [19] that impair very precise positioning. Close locations might share similar fingerprints and, therefore, the typical accuracy of Wi-Fi fingerprints is within a few meters (1–10 m according to [20]). Advanced fingerprint methods report an accuracy of about 1 to 4 m [14,45] but the heterogeneity of testing environments hinders a fair comparison. Even in those cases where the

positioning error is low, there are large errors. For instance, the UMinho system reported an average error of 6.33 m in the 2016 IPIN Competition [21], but large errors were observed: in 5.7% of cases, the positioning error was higher than 20 m and in two cases the error was higher than 40 m.

4. Error Sources in Wi-Fi Fingerprinting

This section is devoted to investigating the origins of positioning errors in Wi-Fi fingerprinting by means of simulated data. The analyses performed in this paper are based on a real scenario, since our long-term objective is to minimize large positioning errors for a particular application in that scenario. However, we consider that most of our findings are useful and can be extrapolated for fingerprinting in general.

The target scenario corresponds to a large laboratory used for polymer research work (PIEP) at the University of Minho, where an indoor positioning system has to be deployed. The space is around 50×20 m wide and more than 8 m high and, in many aspects, is very similar to a factory plant, with large machinery, plenty of metal structures and tools, and some quite large open spaces. Therefore, the considered scenario for simulation is a rectangular area of 50×20 m that resembles the real laboratory. Also, an initial AP distribution with eight Access Points installed near the longer walls has been considered: four of them (APs 1 to 4) placed at 3.9 m above the floor, and the other four (APs 5 to 8) placed at 5.4 m as done in [46].

In the simulations, the default indoor positioning system base parameters are as follows: kNN algorithm with k = 1 (or closest match); the above-mentioned distribution of APs; reference fingerprints in a 1 m grid; and 10 fingerprints per reference point. Simulated base data (RSSI values) are generated with the free space path loss model (see Equation (1)). The selected values for the IPS and the FSPL balance efforts —to generate the radio map and compute the operational fingerprints—, resemble some features of real environments, and they ease the study of the rest of the parameters.

$$FSPL = 20 \cdot \log_{10}(d) + 20 \cdot \log_{10}(f) + adjust constant$$
(1)

where *d* refers to the distance to the AP and *f* corresponds to the signal frequency.

4.1. Quantization of RSSI Values

The Wi-Fi signal is an electromagnetic wave whose intensity is attenuated as it propagates through space. In the optimal case, where the signal is not affected by any external interference, the free space path loss model (see Equation (1)) can be used to calculate the signal attenuation in an optimistic world or scenario. In order to overcome the omnidirectional propagation of the radio signal, multiple APs are used to obtain accurate positioning in 2D and 3D environments as shown in Figure 2, where two naïve examples for three APs are shown. Circles have been used to represent the places where a particular RSSI value is detected for a given AP (red circle for AP_1 , blue circle for AP_2 and green circle for AP_3). In the first example, the operational fingerprint $fp = {rssi_1, rssi_2, rssi_3}$ represents a case where $rssi_1 >> rssi_2 >> rssi_3$, whereas the RSSI values are similar ($rssi_1 \approx rssi_2 \approx rssi_3$) in the second example. In the two examples, a "perfect" unique positioning estimation is provided in the intersection of the three circles.

Despite the signal strength being continuous, the devices often report the detected signal level as integer *dBm* values without decimals. It is worth noting that this quantization only removes the decimal part or rounds to the nearest integer, and does not convert the signal strength to a percentage or quality value. Figure 3 shows some examples of the RSSI according to FSPL. Figure 3a corresponds to the RSSI values from distances comprised between 0 and 120 m from the AP calculated with Equation (1); similarly, Figure 3b shows the values for the quantized world where the RSSI values are rounded to the closest integer value. This quantization step might add uncertainty to the RSSI values.



Figure 2. Illustrative examples of the areas where a fingerprint can be placed in the optimistic world. (a) Fingerprint placed near to AP_1 . (b) Fingerprint equidistant to all APs.



Figure 3. Received signal strength intensities (RSSI) against distance for optimistic (**a**) and quantized (**b**) worlds. (**c**,**d**) Excerpts of (**a**). (**e**) Excerpt of (**b**).

RSSI values are more representative when they are close to the emitter in the optimistic world with neither quantization nor noise; e.g., a decrease of just 1 dBm corresponds to a displacement of \approx 13 cm (from -40.5 dBm to -41.5 dBm) but it is \approx 11.5 m (from -79.5 dBm to -80.5 dBm) according to Figure 3c,d, respectively. This increment of distance, associated with a decrease of 1 dBm, linearly depends on the distance to the AP as can be seen in Figure 4a.

In the quantized world, the RSSI values are also more representative when they are close to the emitter but rounding them to the closest integer value has a collateral effect: the same RSSI value is seen at a range of distances from the AP (see Figure 3b). Now, the decrease of just 1 dBm has not attached a unique point after rounding the values provided by the FSPL (quantization); e.g., the decrease of 1 dBm from -80 dBm to -81 dBm might be 0 m ($\Delta dist_1$ in Figure 3e), ≈ 11.52 m ($\Delta dist_2$ in Figure 3e), ≈ 10.2 m ($\Delta dist_3$ in Figure 3e) or ≈ 21.7 m ($\Delta dist_4$ in Figure 3e). In the best case, the distance (e.g., $\Delta dist_1$ and $\Delta dist_3$) is lower bounded by 0 m and upper bounded by the distance calculated in the optimistic world. In the worst case, the distance (e.g., $\Delta dist_2$ and $\Delta dist_4$) is lower bounded by the distance (see Figure 4b,c). The real RSSI value is shown as a dashed blue line in Figure 3e.



Figure 4. Importance of Δ -1 dBm in the. optimistic (**a**) and quantized (**b**,**c**) worlds

Table 1 is introduced to analyse how quantization might affect fingerprinting. The table shows five representative examples, or cases, at three different base distances to the AP. The RSSI values are compared in the optimistic and quantized worlds.

Dist ₁	RSSI1	ORSSI:	Dista	RSSI	ORSSI	δRSSI	δORSSI	δDist [m]	Est Dist[m]
	Roon	QROON	Dist	Rool	QROOL	01000	UQ1001	(Dist [iii]	Lot. Dist [m]
1.0593	-40.50	-41	1.0594	-40.50	-41	≈ 0	0	< 0.001	0
1.0593	-40.50	-41	1.0592	-40.499	-40	≈ 0	1	< 0.001	≈ 0.13
1.0593	-40.50	-41	1.1885	-41.499	-41	≈ 1	0	≈ 0.13	0
1.0593	-40.50	-41	1.1886	-41.50	-42	≈ 1	1	≈ 0.13	≈ 0.13
1.0592	-40.49	-40	1.1886	-41.50	-42	≈ 1	2	≈ 0.13	≈ 0.26
59.5663	-75.50	-76	59.5664	-75.5	-76	≈ 0	0	<1 mm	0
59.5663	-75.50	-76	59.5662	-75.499	-75	≈ 0	1	<1 mm	\approx 7.27
59.5663	-75.50	-76	66.8343	-76.499	-76	≈ 1	0	\approx 7.27	0
59.5663	-75.50	-76	66.8344	-76.50	-77	≈ 1	1	\approx 7.27	\approx 7.27
59.5662	-75.49	-75	66.8344	-76.50	-77	≈ 1	2	\approx 7.27	≈ 14.54
94.4061	-79.50	-80	94.4062	-79.50	-80	≈ 0	0	<1 mm	0 m
94.4061	-79.50	-80	94.4060	-79.499	-79	≈ 0	1	<1 mm	≈ 11.52
94.4061	-79.50	-80	105.9253	-80.499	-80	≈ 1	0	≈ 11.52	0
94.4061	-79.50	-80	105.9254	-80.50	-81	≈ 1	1	≈ 11.52	≈ 11.52
94.4060	-79.49	-79	105.9254	-80.50	-81	≈ 1	2	≈ 11.52	≈ 23.04

Table 1. Examples of how quantization provides a distance difference of two RSSI with ± 1 dBm higher than expected.

- 1. In the first case, a difference of ≈ 0 dBm in the optimistic world (δ RSSI) corresponds to a difference of 0 dBm in the quantized world (δ QRSSI) as expected. The real and estimated distance between the two points is almost 0 m as expected.
- 2. In the second case, a difference of ≈ 0 dBm in the optimistic world (δ RSSI) corresponds to a difference of 1 dBm in the quantized world (δ QRSSI). Although the real distance between the two points is 0 m, the estimated distance is higher due to quantization: ≈ 0.13 m, ≈ 7.72 m and ≈ 11.52 m for the three base distances shown in the table.
- 3. In the third case, a difference of ≈ 1 dBm in the optimistic world (δ RSSI) corresponds to a difference of 0 dBm in the quantized world (δ QRSSI). Although the estimated distance between the two points is 0 m due to quantization, the real distance is higher: ≈ 0.13 m, ≈ 7.72 m and ≈ 11.52 m for the three base distances shown in the table.
- 4. In the fourth case, a difference of ≈ 1 dBm in the optimistic world (δ RSSI) corresponds to a difference of 1 dBm in the quantized world (δ QRSSI). The real and estimated distances between

the two points match: ≈ 0.13 m, ≈ 7.72 m and ≈ 11.52 m for the three base distances shown in the table.

5. In the fifth case, a difference of ≈ 1 dBm in the optimistic world (δ RSSI) corresponds to a difference of 2 dBm in the quantized world (δ QRSSI). The estimated distances between the two points doubles the real distance between them: ≈ 0.26 m, ≈ 15.54 m and ≈ 23.04 m for the three base distances shown in the table.

It is worth noting that the difference between the real and estimated distances depends on the distance with respect to the AP in all cases.

The five cases shown in Table 1 demonstrate that the δ QRSSI values have an uncertainty of ± 1 dBm, e.g., a difference of RSSI in the real world of 1 dBm might correspond to a difference of RSSI in the quantized world of 0, 1 or 2 dBm. In summary, rounding the RSSI values has introduced some uncertainty which might be considered a source of error. On the one hand, a difference of 0 dBm, which corresponds to a perfect match in the optimistic world, may have attached a large positioning error. On the other hand, a difference of 1 dBm, which might correspond to two separated positions, might correspond to a perfect match. If the RSSI comparison is done close to the antenna, the difference between the real and estimated distances is low (less than 50 cm). However, this difference becomes higher and higher as the distance to the antenna increases (about 7–14 m for points placed at 60 m and 11–23 m for points placed at 95 m) according to the FSPL equation.

The uncertainty introduced by the quantization of RSSI values and the omnidirectional signal propagation are partially overcome by the use of RSSI values from multiple APs. Figure 5 shows two naïve examples of fingerprinting for three APs. Rings have been used to represent the area where a particular RSSI value is detected for a given AP since quantization is present (red ring for AP_1 , blue ring for AP_2 and green ring for AP_3).



Figure 5. Illustrative examples of the areas where a fingerprint can be placed in the quantized world. (a) Fingerprint placed near *AP*₁. (b) Fingerprint equidistant to all APs.

In Figure 5a, the operational fingerprint $fp = \{rssi_1, rssi_2, rssi_3\}$ represents a case where $rssi_1 >> rssi_2 >> rssi_3$. It represents the case where the position is near to just one of the APs. In this first example, the ring area for AP_1 is much smaller than the ring area for AP_3 , since the RSSI value is much higher (stronger). Therefore, the ring area for AP_1 has the lowest radius (which is related to the FSPL) and width (which is related to the uncertainty due to the distance to the AP), which is in line with the reliability of the estimated distance found in [47]. The small area where this fingerprint can be detected is shown on the right side of the figure. This area is small because of the proximity to one of the APs.

In Figure 5b, the operational fingerprint $fp = \{rssi_1, rssi_2, rssi_3\}$ represents a case where $rssi_1 \approx rssi_2 \approx rssi_3$. It represents the case where the distance of all APs to the unknown position is similar. In this second example, the size of the three ring areas is similar. The area where this fingerprint can be detected is higher than for the first example because the aggregated uncertainty considering each individual AP is higher for this particular fingerprint. It is important to note that the same AP distribution has been considered in both examples.

Contrary to the optimistic world where a fingerprint is unique, the same exact fingerprint can be placed within an area (intersection of the three rings) in the quantized world (without noise). This area, considering its size and shape, highly depends on the place where the fingerprint was taken and the AP distribution.

4.1.1. Quantization in the Evaluation Scenario

The previous subsection introduced the uncertainty due to quantization in the determination of the distance to a single AP. In order to analyse the quantization of RSSI values as a source of positioning errors in a realistic indoor environment, some different simulations were carried out. The following default indoor positioning system parameters previously mentioned were applied: Deterministic k-NN algorithm with k = 1 as an indoor positioning system; the 50 × 20 m area resembling the PIEP laboratory, the AP distribution with eight APs previously mentioned; and reference fingerprints in a 1 m grid. For the quantized world, only one fingerprint per reference point was generated since fingerprints do not contain any noise (generating the same exact fingerprint multiple times makes no sense in this case). In the radio map, the reference points were placed at the grid intersection points (totalling $1071 - 51 \times 21$ - reference points), whereas each grid cell contained an operational fingerprint in a random position (totalling $1000 - 50 \times 20$ - evaluation points). This procedure was repeated 100 times in order to have a more representative averaged mean error and maximum error. This corresponds to the base simulation setup followed in all the experiments carried out in this section.

Table 2 shows the accuracy of the 1-NN algorithm in the optimistic world (no quantization and no noise) and in the quantized world (no noise). The mean error corresponds to the average of the mean positioning error over the 100 repetitions, the maximum error corresponds to the average of the maximum positioning error over the 100 repetitions, the percentage of cases above the Maximum Expected Error (MEE) corresponds to the number of operational fingerprints whose error was higher than the MEE. Assuming that the 1-NN algorithm returns the best match, the MEE should never be higher than 0.707 m ($\sqrt{0.5^2 + 0.5^2}$) for the 1 m grid. This distance corresponds to the highest distance between an operational point and its best match (closest fingerprint in the real-world space).

	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
no quantization with quantization	$\begin{array}{c} 0.389 \pm 0.015 \\ 0.493 \pm 0.027 \end{array}$	$\begin{array}{c} 0.933 \pm 0.009 \\ 1.755 \pm 0.015 \end{array}$	0.707 0.707	1.367 20.222

Table 2. Analysis of quantization for Wi-Fi fingerprinting using 1-NN.

According to the simulation results shown in Table 2, quantization has a low impact on the mean positioning error, which increases less than 11 cm on average. However, the maximum reported error is almost doubled and the percentage of cases above the maximum expected error (0.707 m for a 1 m grid) increases from 1.367% to 20.222% of cases on average. The increase in the maximum error and the percentage of cases above the maximum expected error shows that quantization has a significant impact on the indoor positioning system since it has introduced large errors which were not present in the optimistic world.

In order to better explore the impact of quantization, the results are also graphically shown in Figure 6. In the figure, the mean positioning error, the maximum positioning error and the percentage of cases where the error was above 0.707 m are shown for each cell of the environment in the optimistic and quantized worlds. These results have been calculated after repeating the simulations 100 times, i.e., each cell in the scatter plot shows the metric value that has been calculated using the positioning error over the 100 evaluation points attached to the cell, one for each of the 100 simulations.



Figure 6. Graphical results for analysing the impact of quantization. (**a**,**b**) mean positioning error per cell in the optimistic and quantized worlds respectively; (**c**,**d**) maximum positioning error per cell in the optimistic and quantized worlds respectively; (**e**,**f**) percentage of cases where the error was higher than MEE per cell in the optimistic and quantized worlds respectively; (**g**,**h**) percentage of cases where the error was higher the error was higher than twice the MEE per cell in the optimistic and quantized worlds respectively.

For the optimistic world, Figure 6a shows that the mean positioning error is comprised of between 0.35 m and 0.4 m in most of the cells. Only a few cells, located near the corners, provide higher mean positioning errors of around 0.5 m. Similarly, the maximum positioning error per cell (Figure 6c) is comprised of between 0.6 m and 0.75 m in most of the cases. The maximum values (above 1 m) are reached near the corners. Also, the percentage of cases where the error was higher than 0.707 m (Figure 6e) is notably higher in the cells located near the corners with a percentage close to 25%, whereas it is lower than 5% in the rest of the cells. In the optimistic world, the error is never higher than twice the MEE (Figure 6g).

When quantization is introduced, the mean positioning error per cell (Figure 6b) increases with respect to the optimistic world, and the locations of high values are more scattered (Figure 6d). This is also the case for the maximum positioning error per cell and the percentage of cases where the error has been higher than 0.707 m (Figure 6f). Positioning errors higher than 1.414 (twice the maximum expected error) appear and the percentage of these errors is between 25% and 40% (Figure 6h) in some cells.

Figure 7 graphically shows the following fingerprint statistics for each reference point in the optimistic and quantized worlds: Uniqueness (number of reference fingerprints with exactly the

same RSSI values) and mean/maximum/median/minimum RSSI value for each reference fingerprint. According to the figure, the reference fingerprints are unique in the optimistic world, i.e., given any two reference fingerprints, the RSSI vectors never completely match. However, there are some cases where the fingerprints are not unique in the quantized world. In a significant number of cases (green cells in Figure 7b), two fingerprints have exactly the same RSSI values. In four cases (red cells in Figure 7b), three fingerprints have exactly the same RSSI values. The other statistics are similar in the optimistic world and quantized world, with those in the optimistic world being more smoothed.



Figure 7. RSSI and fingerprint statistics in the optimistic world (left images) and quantized world (right images). (**a**,**b**) uniqueness of reference fingerprints; (**c**,**d**) mean RSSI value of reference fingerprints; (**e**,**f**) maximum RSSI value of reference fingerprints; (**g**,**h**) median RSSI value of reference fingerprints; (**i**,**j**) minimum RSSI value of reference fingerprints.

There is no doubt that quantization has increased the positioning errors. If the statistics of the RSSI values were analysed, it could be observed that most of the large errors are located near the areas where the reference fingerprints were similar and they were not unique.

4.2. Noise of RSSI Values

The electro-magnetic signals are affected by many features and elements present in the environment, which introduce reflection, refraction, absorption and diffraction (among others) in the signal. Therefore, the signal strength fluctuates when measured over time at a given position.

The addition of white Gaussian noise is the usual starting point for understanding basic performance relationships in the study of communication systems [48]. Although this assumption is not always true, it significantly simplifies the computations with little performance loss [34]. According to [36], most of RSSI distributions (70% approx.) are often left-skewed, despite the normal distribution being usually used. Although consensus about the best model to fit the RSSI data distributions has not been reached [49], most authors agree that the RSSI histograms resemble a Gaussian distribution in most of the cases. Therefore, in the rest of this paper, we assume that noise in the RSSI values can be modelled by a Gaussian distribution with null mean and standard deviation (σ) between 1 and 5.

Figure 8 shows some examples of the RSSI according to the distance to the AP using the path loss with Gaussian noise ($\sigma = 2$). In particular, Figure 8a shows the signal strength calculated with the path loss model (blue line) plus the bounds corresponding to the 68–95–99 rule (i.e., the 68.27%, 95.45% and 99.73% of noisy RSSI values fall within the first, second and third bounds respectively). Similarly, Figure 8c shows the path loss equation when noise is present ($\sigma = 2$) and the RSSI values are quantized.

Figure 8b shows an example of the uncertainty introduced by noise. First, $\Delta dist_0$ shows the range of distances to the AP where the same RSSI, -71 dBm, can be obtained in 68.27% of cases for the path loss model with $\sigma = 2$. Although most of the injected noise is close to the zero-mean according to the Gaussian distribution, there might be a few outliers whose error is considerably high (more than three times σ) as occurs in a real scenario (see an outlier example in Figure 8d at 49.5 m from the AP). Second, $\Delta dist_1$ shows that two measures taken at close positions might have a large difference in RSSI (≈ 9 dBm in the example) but the same difference in RSSI might correspond to two separated positions as shown in $\Delta dist_2$. Finally, the same RSSI can be detected at two separated positions as shown in $\Delta dist_3$. Figure 8d shows an example of the uncertainty in the realistic simulated world, where quantization and noise are both present. In general, Figure 8 shows that the noise present in the radio signal adds more uncertainty in order to estimate a position.

In the rest of this paper, some simulations will be carried out in the optimistic world (no quantization, no noise), in the quantized world (quantization, no noise), and in the realistic word (quantization, noise). The base of all of them is the path loss equation introduced in Equation (1); quantization is just rounding the signal strength to the nearest integer value, and the noise is injected using a normal distribution with zero mean and $\sigma = 2$ (other values of σ will be occasionally explored). The configuration for the realistic world has been used in many other previous works and we consider that it is a valid configuration to explore the sources of errors, which are inherited from the methods and assumptions in deterministic fingerprinting.



Figure 8. Received signal strength intensities (RSSI) against distance for the realistic noisy world ($\sigma = 2$) without quantization (**a**) and with quantization (**c**). (**b**) Excerpt of (**a**). (**d**) Excerpt of (**c**).

4.3. The Grid Size

In order to analyse the grid size as a source of positioning errors, some simulations were carried out. In particular, the following different grid sizes were considered: 10 m, 5 m, 2 m, 1 m, 0.5 m, 0.2 m and 0.1 m. In order to have a more comprehensive study, the analysis considers the optimistic (without quantization and noise), quantized (with quantization and without noise) and realistic (with quantization and noise) worlds.

4.3.1. The Optimistic World

For the optimistic world, the results of applying different grid sizes are shown in Table 3.

Table 3. Analysis of grid size for Wi-Fi fingerprinting in the optimistic world using 1-NN: Mean positioning error; Maximum positioning Error; Maximum expected error (MEE) and percentage of cases above the MEE.

Grid	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
10.0 m	3.943 ± 0.156	6.292 ± 0.080	7.071	2.003
5.0 m	1.975 ± 0.077	3.607 ± 0.035	3.536	1.875
2.0 m	0.776 ± 0.030	1.600 ± 0.012	1.414	1.416
1.0 m	0.389 ± 0.015	0.933 ± 0.009	0.707	1.367
0.5 m	0.194 ± 0.008	0.525 ± 0.004	0.354	1.340
0.2 m	0.078 ± 0.003	0.227 ± 0.001	0.141	1.296
0.1 m	0.039 ± 0.002	0.119 ± 0.000	0.071	1.279

According to Table 3, the positioning error depends on the grid size when fingerprinting is applied in the optimistic world without quantization and noise. The lower the grid size, the lower the mean positioning error, which is the common assumption in fingerprinting. However, this table shows an interesting finding, even in the ideal world without noise and quantization, there are a few cases (around 1% and 2% depending on the grid size) where the positioning errors are larger than expected. Another interesting result, is that the average of the maximum error over the 100 simulations is lower than the MEE threshold for the grid size of 10 m. On the other hand, the maximum positioning error almost doubles the MEE for the grid size of 10 cm.

In order to better analyse where the large error occurs, Figure 9 graphically shows the mean positioning error per grid cell and the percentage of cases where the positioning error was higher than the maximum expected error; only the 0.5 m and 1 m grid have been considered.



Figure 9. Graphical results for analysing the grid size in the optimistic world. (**a**,**c**) mean positioning error per cell in the optimistic world for a grid size of 0.5 m and 1 m respectively; (**b**,**d**) percentage of cases where the positioning error is above the MEE for a grid size of 0.5 m and 1 m respectively.

The graphical accuracy shown in Figure 9 clearly depicts a trend in the evaluation scenario, as there are some areas (cells) where the mean positioning error is clearly higher than in the rest; i.e., there are areas where the fingerprints are smoothed and the closest fingerprint in the RSSI space might not correspond to the closest fingerprint in the geometric space. Four of these areas correspond to the scenario corners. Moreover, a geometric pattern is clearly observed as being attached to the antenna position in Figure 9b,d, where the percentage of cases above the maximum expected error is shown.

4.3.2. The Quantized World

Table 4 shows the results for the different grid sizes in the quantized world; the results for the optimistic world were introduced in a similar fashion.

The positioning error also depends on the grid size in the quantized world. However, there is a lower bound in the mean and maximum positioning errors for grid sizes below 1 m. Also, the percentage of cases above the MEE increases for grid sizes below 1 m. Another interesting finding is that the maximum error is very large (more than twice the MEE) for grid sizes lower than 1 m, which indicates that the estimated position is outside the cell delimited by the four surrounding reference points. 0.1 m

 0.312 ± 0.017

Grid	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
10.0 m	3.706 ± 0.157	6.083 ± 0.070	7.071	0.800
5.0 m	1.825 ± 0.079	3.557 ± 0.040	3.536	1.675
2.0 m	0.787 ± 0.039	2.039 ± 0.019	1.414	6.924
1.0 m	0.493 ± 0.027	1.755 ± 0.015	0.707	20.222
0.5 m	0.370 ± 0.021	1.617 ± 0.011	0.354	47.243
0.2 m	0.318 ± 0.019	1.418 ± 0.004	0 141	82.348

0.071

94.240

Table 4. Analysis of grid size for Wi-Fi fingerprinting in the quantized world using 1-NN: Mean positioning error; Maximum positioning Error; Maximum expected error (MEE) and percentage of cases above the MEE.

Figure 10 graphically shows the mean positioning error per grid cell and the percentage of cases where the positioning error was higher than the maximum expected error for the 0.5 m and 1 m grids.

 1.465 ± 0.004



Figure 10. Graphical results for analysing the grid size in the quantized world. (a,c) mean positioning error per cell in the quantized world for a grid size of 0.5 m and 1 m respectively; (b,d) percentage of cases above the MEE in the quantized world for a grid size of 0.5 m and 1 m respectively.

In the quantized world, there are still some areas where the mean positioning error is clearly higher than in the rest (see Figure 10). When compared to the same figures for the optimistic world (see Figure 9), it can be observed that the number of cases above the maximum expected error has increased. According to the results and the analysis presented in this section, quantization is a phenomenon present in fingerprinting which renders more difficult the task required to differentiate fingerprints collected at nearby positions.

4.3.3. The Realistic Noisy World

Finally, the realistic noisy world is analysed. Table 5 shows the results for the different grid sizes in the quantized world; the results for the optimistic and quantized worlds were introduced in a similar fashion. In this table, five different levels of noise (σ value for the Gaussian distribution) are considered.

In the realistic noisy world (with quantization and noise), the positioning error depends on the grid size and variance of the RSSI values. On the one hand, the lower the grid size, the lower the mean positioning error. On the other hand, the lower the variability of RSSI readings, the lower the positioning error. However, the combination of grid size and signal variability leads to an interesting finding: the grid size has little impact on the mean accuracy of the lower grid sizes as the signal variability becomes high, e.g., the accuracy obtained using a 2 m grid is very similar to the accuracy obtained using a 1 m grid for any σ value, but the number of reference points is four times lower in the former case (2 m grid).

Noise	Grid	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
	10.0 m	4.086 ± 0.180	6.968 ± 0.115	7.071	5.500
	5.0 m	2.389 ± 0.122	5.508 ± 0.101	3.536	17.250
$\sigma = 1$	2.0 m	1.852 ± 0.106	6.121 ± 0.084	1.414	60.872
	1.0 m	1.728 ± 0.099	6.255 ± 0.075	0.707	86.154
	0.5 m	1.647 ± 0.095	6.781 ± 0.054	0.354	95.973
	10.0 m	4.849 ± 0.245	9.045 ± 0.169	7.071	19.100
	5.0 m	3.761 ± 0.205	9.192 ± 0.156	3.536	49.325
$\sigma = 2$	2.0 m	3.370 ± 0.192	10.870 ± 0.159	1.414	85.744
	1.0 m	3.218 ± 0.184	11.764 ± 0.111	0.707	95.587
	0.5 m	3.102 ± 0.178	12.512 ± 0.098	0.354	98.826
	10.0 m	5.996 ± 0.328	11.667 ± 0.318	7.071	32.300
	5.0 m	5.254 ± 0.298	13.373 ± 0.238	3.536	68.050
$\sigma = 3$	2.0 m	4.827 ± 0.282	16.322 ± 0.277	1.414	92.312
	1.0 m	4.610 ± 0.268	18.498 ± 0.294	0.707	97.787
	0.5 m	4.492 ± 0.263	21.336 ± 0.412	0.354	99.395
	10.0 m	7.717 ± 0.425	15.248 ± 0.416	7.071	52.000
	5.0 m	6.826 ± 0.416	18.884 ± 0.569	3.536	79.425
$\sigma = 4$	2.0 m	6.242 ± 0.379	23.570 ± 0.545	1.414	94.924
	1.0 m	6.010 ± 0.365	27.152 ± 0.575	0.707	98.633
	0.5 m	5.855 ± 0.356	32.759 ± 0.656	0.354	99.622
	10.0 m	9.353 ± 0.567	19.674 ± 0.671	7.071	60.400
	5.0 m	8.308 ± 0.517	23.980 ± 0.655	3.536	83.750
$\sigma = 5$	2.0 m	7.676 ± 0.481	30.767 ± 0.745	1.414	96.768
	1.0 m	7.376 ± 0.468	37.907 ± 0.678	0.707	99.004
	0.5 m	7.191 ± 0.456	43.149 ± 0.509	0.354	99.738

Table 5. Analysis of grid size for Wi-Fi fingerprinting in the noisy world using 1-NN: Mean positioning error; Maximum positioning Error; Maximum expected error (MEE) and percentage of cases above MEE.

A second interesting finding can be observed in the table: the maximum error and the percentage of cases where the error was higher than the maximum expected error increase as the grid size decreases (except for just one case with grid = 10 m and $\sigma = 1$), which is consistent with the results shown for the quantized world. Also, both increase as the injected noise (σ) increases, i.e., the probability of obtaining an error higher than the MEE depends on the signal variability and it inversely depends on the grid size according to the simulations carried out.

It is important to remark that, contrary to the common rationale, decreasing the grid size does not guarantee a corresponding decrease in the mean positioning error. In fact, the probability of having large positioning errors increases when the grid size is lower than 2 m, which might be due to the high density of fingerprints.

Figure 11 graphically shows the accuracy (as mean and maximum positioning errors) of the 1 m grid configuration and different values of RSSI signal variance (from 0 to 5). This figure clearly shows that the error is not uniform in the grid.

Observing the mean positioning errors depicted in Figure 11, a pattern arises. The areas with low mean positioning error correspond to the location of antennas and their surroundings. The cells providing a large mean positioning error are located in three main parts of the scenario: the horizontal line that divides the scenario into two symmetrical parts, the vertical line that divides the scenario into two symmetrical parts, the vertical line that divides the scenario into two symmetrical parts and the cells located in the periphery (especially on the top, left and right sides). The cells close to the horizontal and vertical lines have the lowest maximum RSSI value (see Figure 7e,f), whereas the cells located in the periphery have two interesting features: (1) they are not surrounded by Wi-Fi APs and (2) they are not uniformly surrounded by other reference points because there are not any reference points outside the scenario (i.e., there are less contiguous cells and reference points).

ing Error per

(a) Mean Positioning Error per Cell [m]

(i) Mean Positioning Error per Cell [m]

(k)



(j) ng Error

(1)

er Cell [m



D.8

0.7

0.6 0.5

0.4

Figure 11. Graphical results (mean and maximum positioning errors over 100 runs) for analysing the grid size in the noisy world. (**a**,**b**) results for $\sigma = 0$ (no noise); (**c**,**d**) results for $\sigma = 1$; (**e**,**f**) results for $\sigma = 2$; (**g**,**h**) results for $\sigma = 3$; (**i**,**j**) results for $\sigma = 4$; (**k**,**l**) results for $\sigma = 5$.

20

15 10

4.3.4. General Discussion about the Grid Size

The previous observed facts about the grid size indicate that the regular division of the environment might not be the best strategy to map an environment since the signal propagation model, the distribution of antennas and other features attached to the signal propagation have not been considered. Even in the optimistic world, it can be seen that the mean and maximum positioning errors are not uniform in all the cells that compose the environment. For instance, the corners in the testing scenario seem more difficult to differentiate.

Some well-established assumptions are not totally true as complexity is added to the operational scenario (quantization and noise), e.g, *"the positioning error decreases as the grid size decreases"*. Small grids are associated with an overall low mean positioning error, but at the expense of having higher maximum errors and more individual cases where the positioning error is higher than the maximum expected error for the corresponding grid size; e.g., in less than 5% of cases, the error was lower than 0.7 m (the maximum expected error for a 1 m grid) for σ equal or higher than 2 (moderate–high variability in RSSI values), but in less than 1% of cases the error was lower than 0.35 m (the MEE for a 0.5 m grid) for σ equal or higher than 3 (moderate–high variability in RSSI values). Also, the accuracy reported with the 2 m grid is similar to the accuracy provided with the 0.5 m grid according to the simulated results.

Furthermore, the simple grid strategy that is usually applied in fingerprinting might be considered a source of positioning error since the RSSI values might be wrongly treated as unitless numbers, as stated in [28]; i.e., the environment is divided according to an arbitrary geometric rule but the physical underpinning of radio signal propagation is ignored when the radio map is generated.

4.4. The Density of APs

This section focuses on the number and distribution of APs as a source of positioning errors. The analysis is also done through a comprehensive simulation.

4.4.1. Theoretical Bounds of AP Density

First, the optimal number of APs was established by simulation. The optimistic, quantized and realistic noisy world scenarios were all considered to analyse the effect of AP density on positioning accuracy in depth. Figure 12a shows a plot with the minimum (green), mean (black) and maximum (red) positioning accuracy by using 1 to 100 APs randomly (uniform distribution) distributed in the scenario; the gray area corresponds to the standard deviation of the mean value of the 100 simulations. In order to avoid slanted results, the random AP distribution is different in each of the 100 simulation repetitions. In the figure, the accuracy corresponds to the positioning error in the optimistic world without quantization and noise. Similarly, Figure 12b,c show the same statistics for the quantized and real world (with $\sigma = 2$).

In Figure 12, it can be observed that there is a lower bound in the positioning accuracy, where adding more APs does not have a huge impact on the accuracy. In the optimistic word, an impressive accuracy is obtained with just three APs if they are placed in the optimal place (minimum error, green line). Adding more APs to the scenario reduces the risk of having a large mean positioning error but there is not a significant improvement in the mean accuracy. In the quantized world, an impressive mean accuracy can also be achieved but the number of APs required is slightly higher (6 APs). In the realistic noisy world (with noise $\sigma = 2$), where this analysis is crucial, the lower bound is about 25–30 APs and 2 m of mean positioning error, i.e., in order to provide competitive accuracy, at least 0.025 APs per m² are required. If the APs are uniformly distributed to cover this scenario without overlappings, this means that the distance of any operational fingerprint to the closest AP should be ≈ 3.6 m or lower ($\pi \cdot r^2 <= \frac{50m \cdot 20m}{25APs}$) considering a 2D scenario. This threshold distance is quite far from the distances to the closest AP in the real scenario, where the distance of 20% of the reference points to the closest AP is more than 8 m. Although the expected mean positioning error

with 100 APs (0.1 APs per m²) is about 1.5 m, installing such a quantity of APs is not feasible in a real deployment. Theoretically, the positioning error might be reduced by adding hundreds of APs, but the improvement is lower bounded by a mean positioning error of about \approx 1.1 m and such deployments are not practical at all.



Figure 12. Impact of the number of APs in the optimistic (**a**); quantized (**b**) and realistic noisy (**c**) worlds. Black lines correspond to the average accuracy over the 100 simulations, green lines correspond to min. accuracy, and the red lines correspond to the max. accuracy.

4.4.2. Impact of AP Distribution

Regarding the AP as a source of error, the AP distribution is also studied in this paper. In particular, we focus on the scenario with eight APs as in the real evaluation scenario where we aim to develop a Wi-Fi-based indoor positioning system. Five alternative distributions (see Table 6) have been considered to show that the AP distribution has a direct impact on the distribution of large errors.

AP Distribution	AP 1/AP 5	AP 2/AP 6	AP 3/AP 7	AP 4/AP 8
Configuration 1	[625,500,390]	[1875,500,390]	[3150,500,390]	[4375,500,390]
	[625,1500,540]	[1875,1500,540]	[3150,1500,540]	[4375,1500,540]
Configuration 2	[0,0,390]	[1666,666,390]	[3333,666,390]	[5000,0,390]
	[000,2000,540]	[1666,1333,540]	[3333,1333,540]	[5000,2000,540]
Configuration 3	[000,666,390]	[1666,0,390]	[3333,0,390]	[5000,666,390]
	[000,1333,540]	[1666,2000,540]	[3333,2000,540]	[5000,1333,540]
Configuration 4	[500,100,390]	[1750,100,390]	[3250,100,390]	[4500,100,390]
	[500,1900,540]	[1750,1900,540]	[3250,1900,540]	[4500,1900,540]
Configuration 5	[313,1000,390]	[938,1000,390]	[1563,1000,390]	[2188,1000,390]
	[2813,1000,390]	[3438,1000,390]	[4063,1000,390]	[4688,1000,390]

Table 6. Location [x,y,z] of the 8 APs.

The first configuration is similar to the real AP deployment, but the APs are uniformly distributed to avoid large distances between the APs located in the top and bottom part of the scenario. In the second configuration, the APs are located in the diagonals of the squared scenario. The third configuration is a variant of the previous one. In the fourth configuration, the APs are located at the original position with a random displacement of 1 m radius to avoid symmetries. In the fifth configuration, all the APs are uniformly distributed at the horizontal line that divides the scenario into two parts. The location of the APs can be inferred from the maximum RSSI value shown in Figure 7e,f.

Table 7 shows the results for the six studied AP configurations, the original one of our real scenario and the five proposed configurations. The results show that the first configuration provides the lowest mean error over the 100 repetitions. However, this accuracy is just a number and the error distribution over the environmental area cannot be extrapolated because it is an average value.

AP Dist	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
Orig	3.237 ± 1.860	12.560	0.707	95.540
Alt 1	3.169 ± 1.962	18.609	0.707	95.050
Alt 2	3.580 ± 2.364	20.622	0.707	95.850
Alt 3	3.433 ± 2.059	17.337	0.707	95.900
Alt 4	3.251 ± 1.868	14.333	0.707	95.470
Alt 5	6.140 ± 5.134	23.195	0.707	96.790

Table 7. Analysis of AP distribution for Wi-Fi fingerprinting using 1-NN.

Figure 13 graphically shows the mean and maximum positioning errors per cell, whereas Figure 14 shows the RSSI statistics (mean and max. RSSI value) for one of the 100 runs. The figures demonstrate how positioning errors are distributed and why large errors occur. Figure 13 clearly shows that the error distribution depends on the locations of the APs (which can be inferred from the maximum RSSI values shown in Figure 14). In the original scenario, with the AP distribution that matches our real environment, the central part of the environment provides the maximum mean positioning error, which is lower than 4 m. In the first alternative, there are many cells located in the left and right sides of the environment which provide a high mean positioning error of about 4.5 m and 5 m, and the maximum error is always lower than 20 m. Similarly, the second and third alternatives show different mean positioning error is higher than 20 m (especially in the second alternative). The fourth alternative shows no significant difference with respect to the original AP deployment

n the last scenario, the effects of symmetry are depicte

since the differences in AP location are small. In the last scenario, the effects of symmetry are depicted. Since all the APs are located in a row, it is not possible to determine in which side the fingerprint is located and the errors are quite high in the top and bottom parts of the scenario, where the mean errors are about 15 m, and the maximum errors are about 25 m. To sum up, the positioning error depends on the AP distribution.



Figure 13. Graphical results (mean and maximum positioning errors over 100 runs) for analysing the AP distribution in the noisy world. (**a**,**b**) original AP distribution; (**c**,**d**) alternative AP distribution 1; (**e**,**f**) alternative AP distribution 2; (**g**,**h**) alternative AP distribution 3; (**i**,**j**) alternative AP distribution 4; (**k**,**l**) alternative AP distribution 5.



Figure 14. RSSI Statistics (mean and maximum RSSI value) for analysing the AP distribution in the noisy world ($\sigma = 2$). (**a**,**b**) original AP distribution; (**c**,**d**) alternative AP distribution 1; (**e**,**f**) alternative AP distribution 2; (**g**,**h**) alternative AP distribution 3; (**i**,**j**) alternative AP distribution 4; (**k**,**l**) alternative AP distribution 5.

When the mean and maximum positioning errors are compared to the statistics of the fingerprints (mean and maximum RSSI value), a pattern again arises. The cells that have a low mean positioning error correspond to the cells which are close to an AP. Moreover, the probability of having a large error is higher in those cells where the mean RSSI value is low.

The results shown in Figure 13 are summarized in Figure 15, where the CDF of the mean positioning error per cell (100 repetitions) is shown. According to these two plots, the original AP distribution seems to be the best one followed by alternative 4 and alternative 1. Although alternative 1 provides better mean accuracy than the original distribution for more than 75% of cells, it provides worse maximum accuracy than the original distribution.



Figure 15. Cumulative distribution of the mean (**a**) and maximum (**b**) positioning error (CDF) for six different AP distributions. Dashed vertical lines indicate the average error, whereas dotted vertical lines indicate the highest error.

4.4.3. Presence of Virtual APs

Nowadays, it is common for the same AP to provide different Wi-Fi networks. The reasons for providing multiple networks are diverse: covering different frequencies and channels (2.4 GHz and 5.2 GHz) to provide Internet access to multiple people; providing access to different networks depending on the person's profile (client, staff, maintenance in a mall); among many others.

Since the AP infrastructure is unknown in some Wi-Fi deployments, the detected networks (RSSI values) are commonly considered as independent networks. In this section, we will analyse the presence of virtual APs as a source of error, but also as a mechanism to improve the accuracy of indoor positioning systems.

For this experiment, it has been considered that each AP can emit up to four independent RF signals (Wi-Fi networks), as this virtual AP configuration has been detected at the University of Minho and Universitat Jaume I. Table 8 shows the mean positioning error for the evaluation scenario with the default AP distribution. An AP distribution with 32 APs is analysed for comparison purposes.

The results of Table 8 show that the default configuration (eight APs, single network) provided a mean positioning error of 3.2 m. By increasing the number of APs to 32, uniformly distributed in the scenario, the error decreases to 1.9 m. Also, the maximum positioning error and the percentage of cases where the error was higher than expected are, both, reduced by using the 32 AP configuration. As previously mentioned, the AP density is higher and, therefore, the IPS performs better.

The table also shows the results of eight APs that emit four different networks each. Here, two options are available: generate a concatenated RSSI vector with 32 values (4×8), or generate a vector with eight values by averaging the four RSSI values emitted by each AP ($avg \times 8$). When the four RSSI values emitted by each AP are totally independent, the results are better than with the original eight single-network APs (see rows 3 and 4). Although both alternatives (concatenating and averaging)

are good, the averaged solution provides better results. On the one hand, 32 independent APs are detected when the four RSSI values per AP are concatenated, but the accuracy of the 32 uniformly distributed APs is not reached. On the other hand, averaging the values mitigates the noise present in the environment, so the training and operational fingerprint are less affected by noise and the best accuracy in the table is reached, i.e., it is better to have eight APs emitting multiple independent networks than having 32 APs emitting a single network according to the simulations.

AP Dist	Independent	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
$1 \times 8 \text{ AP}$	-	3.218 ± 0.184	11.764 ± 0.111	0.707	95.587
$1 \times 32 \text{ AP}$	-	1.967 ± 0.102	6.753 ± 0.078	0.707	89.703
$4 \times 8 \text{AP}$	yes	2.160 ± 0.121	7.495 ± 0.067	0.707	91.146
$avg \times 8 AP$	yes	1.815 ± 0.102	6.517 ± 0.065	0.707	87.636
$4 \times 8 \mathrm{AP}$	partial	2.635 ± 0.148	9.321 ± 0.083	0.707	93.794
$avg \times 8 AP$	partial	2.420 ± 0.137	8.653 ± 0.092	0.707	92.836
$4 \times 8 \text{AP}$	no	3.206 ± 0.184	11.710 ± 0.157	0.707	95.640
$avg \times 8 AP$	no	3.209 ± 0.183	11.569 ± 0.114	0.707	95.733

Table 8. Analysis of virtual APs for Wi-Fi fingerprinting in the realistic world using 1-NN.

When the four APs are partially independent (present a dependency of 50%, e.g., the values from the first and second networks are correlated or identical; and the values from the third and fourth networks are also correlated or identical), the use of virtual APs also improves the accuracy of fingerprinting, but the difference with respect to the original eight single-network APs is lower. If the RSSI values of the four networks are totally correlated (the worst possible scenario for an environment with virtual APs), the accuracy is similar to the original eight single-network APs.

Figure 16 shows the mean and maximum positioning error per cell to compare the results provided without virtual APs (8 and 32 APs) and with virtual APs (concatenating them and averaging them). Figure 17 shows the CDFs of these four configurations in the best possible scenario for the virtual APs (the four RSSI values are not correlated).

It seems that having virtual APs benefits fingerprinting since the presence of outlier RSSI measurements might be reduced. However, in most fingerprinting solutions, virtual APs are not exploited and they are considered as independent networks. Not exploiting this information might be considered a source of error since the RSSI readings remain noisy and the density of APs is artificially increased.

4.4.4. General Discussion about the AP Density

First, the AP density might be considered a source of large errors in Wi-Fi fingerprinting. If the number of detected APs is low, the probability of having large errors is higher since there is not enough information to distinguish fingerprints in a scenario.

The distribution of APs is also important. In those areas where the distance to the closest AP is higher than a threshold, the probability of having a large error increases. In our simulated scenario, the threshold value was \approx 3.568 and the largest errors were located in those areas where the distance to the closest APs was larger than this threshold. Hopefully, this source of error can be detected at the operational fingerprint by its maximum RSSI value.

Regarding the presence of virtual APs, they have to be carefully managed. If the virtual APs are considered as independent networks, the AP density is artificially increased and the positioning error is slightly larger than expected, i.e., 32 uniformly distributed APs provide better accuracy than eight uniformly distributed APs emitting four networks. However, the best results are obtained if the knowledge about virtual APs is exploited, i.e., eight averaged uniformly distributed APs provide better accuracy than 32 single-net uniformly distributed APs.



Figure 16. Graphical results (mean and maximum positioning errors over 100 runs) for analysing virtual APs in the noisy world. (**a**,**b**) eight APs, one network per AP; (**c**,**d**) eight APs, four networks per AP (concatenated); (**e**,**f**) eight APs, four networks per AP (average); (**g**,**h**) 32 APs, one network per AP.

(h)

(g)



Figure 17. Cumulative distribution of the mean (**a**) and max. (**b**) positioning error (CDF) for comparison of virtual APs. Dashed and dotted vertical lines stand for average and highest error, respectively.

4.5. Number of Fingerprints

In order to analyse the number of fingerprints, or FP density, as sources of positioning errors, some different simulations were carried out in a similar fashion as introduced in the previous studies.

4.5.1. Theoretical Bounds of Reference FP Density

First, the optimal number of fingerprints was established by simulation. For this fingerprint feature, only the real world scenario was considered to analyse the effect of reference FP density on positioning accuracy in depth. In the optimistic and quantized world, it makes no sense to collect multiple fingerprints on the same place since the same exact fingerprint is replicated.

Figure 18 shows a plot with the minimum (green), mean (black) and maximum (red) positioning accuracy by using 1 to 100 FPs per reference point (gray areas represent the standard deviation of the mean positioning accuracy). The minimum, mean and maximum values are calculated based on the performance over the 100 runs. In the figure, the accuracy corresponds to the positioning error in the realistic noisy world for $\sigma = 2$ (Figure 18a) and $\sigma = 4$ (Figure 18b).



(a)



Figure 18. Impact of the number of reference fingerprints in the realistic world with noise $\sigma = 2$ (a) and $\sigma = 4$ (b). Black lines correspond to the average accuracy over the 100 simulations, green lines correspond to min. accuracy, and the red lines correspond to the max. accuracy.

In Figure 18a, it can be observed that there is a lower bound in the accuracy when $\sigma = 2$. The mean positioning accuracy is about 3.5 m with just one fingerprint per reference point (3.7 m in the worst run)

and 3.1 m with 100 fingerprints per reference point (2.95 m in the best run). With 10 fingerprints per reference point, a common value, the accuracy is about 3.20 m (oscillating between 3.1 and 3.3 in the best and worst runs respectively). It seems that the FP density per reference point has a low impact on the accuracy since the variance of the RSSI signal is 2: the mean positioning error oscillates between \approx 3 m and \approx 3.7 m. However, it is worth mentioning that the operational fingerprint is just one vector of the RSSI values which might contain outlier values due to the noise. For $\sigma = 4$, the trends are similar but the mean positioning error is higher (around 6 m).

To better analyse the distribution of the errors, Figure 19 shows the mean and maximum positioning errors per cell for 1, 10 and 100 fingerprints per reference point. For simplification, only $\sigma = 2$ is considered.



Figure 19. Graphical results (mean and maximum positioning errors over 100 runs) for analysing the AP distribution in the noisy world. (**a**,**b**) one fingerprint per reference point; (**c**,**d**) 10 fingerprints per reference point; (**e**,**f**) 100 fingerprints per reference point.

Figure 19 clearly shows that the mean and maximum positioning errors decrease as the number of fingerprints per reference point increases. Although the position of large errors does not vary, the probability of obtaining such errors decreases. However, the difference in the accuracy between 10 and 100 fingerprints per reference point is marginal. It seems that the density of fingerprints per reference point is a parameter that should be carefully set since a high density of fingerprints per reference point might not drastically reduce large positioning errors and the computational cost might be prohibitive.

4.5.2. Averaging the Reference Dataset

According to the results shown in the previous subsection, it seems that adding more reference fingerprints does not have the desired impact on the IPS accuracy, and the computational cost of k-NN-based indoor positioning systems might significantly increase. However, one can reduce the

number of fingerprints by averaging them, i.e., forming a new fingerprint by averaging the RSSI values provided by *n* fingerprints at the same place.

Figure 20a shows the positioning accuracy by using 1 to 100 FPs, where the values have been averaged into blocks of five non-overlapping fingerprints. In the plot, '100 (individual) fingerprints' corresponds to '20 averaged fingerprints'. Similarly, Figure 20b also shows the positioning accuracy by using 1 to 100 FPs, but the values have been averaged in blocks of 10 non-overlapping fingerprints ('100 (individual) fingerprints' corresponds to '10 averaged fingerprints'). This procedure might reduce the noise and size of the radio map.



(b)

Figure 20. Impact of the number of reference fingerprints in the realistic world with averaging in blocks of 5 (**a**) and averaging in blocks of 10 (**b**). Black lines correspond to the average accuracy over the 100 simulations; green and red lines correspond to min. and max. accuracy, respectively.

In Figure 20a, it can be observed that the lower bound shown in Figure 18 persists. Averaging the reference fingerprints reduces the mean positioning error to about 2.6 m, which is significantly lower than the accuracy shown in Figure 18. With just one averaged fingerprint per reference point (averaging five independent individual fingerprints), the mean positioning accuracy is about 2.8 m (2.9 m and 2.7 m in the worst and best case respectively) and 2.7 m with 20 averaged fingerprints per reference point (2.8 m and 2.6 m in the worst and best case respectively). For the case of 10 individual independent fingerprints per reference point, the error is reduced by \approx 15%, since the accuracy was about 3.20 m before averaging and it is 2.77 m when fingerprint averaging is applied.

In Figure 20b, it can be observed that the accuracy is not highly improved with respect to the results shown in Figure 20a. The reported accuracy and the lower bounds are similar. With just one averaged fingerprint per reference point (averaging 10 independent single fingerprints), the mean positioning accuracy is about 2.75 m (2.9 m and 2.6 m in the worst and best case respectively) and 2.65 m with 10 averaged fingerprints per reference point (2.8 m and 2.45 m in the worst and best case respectively). For 10 individual independent fingerprints per reference point, the accuracy was about 3.20 m before averaging and it is 2.75 m when the fingerprint average is applied.

Increasing the number of reference fingerprints, averaged or not, has a lower bound in the mean and maximum positioning error as can be seen in Figures 18 and 20. Adding more fingerprints has entailed an increase in the computational cost at the training (collection of fingerprints) and operational (estimate position) stages. However, the increase in accuracy might not be as expected. Although the noise in the reference fingerprints is mitigated by averaging, there is a source of positioning error which is not covered: the noise present in the operational fingerprint.

To better analyse the distribution of the errors, Figure 21 shows the mean and maximum positioning errors by using 10 independent single reference fingerprints, two averaged fingerprints (average of five independent single fingerprints) and one averaged fingerprint (average of 10 independent single fingerprints). For simplification, only $\sigma = 2$ is considered.



Figure 21. Graphical results (mean and maximum positioning errors over 100 runs) for analysing the impact of averaging reference fingerprints in the realistic noisy world. (**a**,**b**) 10 fingerprints per reference point without averaging; (**c**,**d**) 10 fingerprints per reference point with averaging in blocks of 5; (**e**,**f**) 10 fingerprints per reference point with averaging in blocks of 10.

Figure 21 shows that averaging reduces the mean and maximum positioning error per cell. Although the positions of large errors do not vary, the probability of obtaining such errors decreases. Also, the difference between averaging in blocks of 5 and 10 fingerprints is low when 10 independent single fingerprints were collected at the training stage. It is worth noting that the computational costs at the operational stage are approximately 10 times lower in the last case (averaging in blocks of 10 fingerprints) than in the first case (no fingerprint average).

4.5.3. Averaging the Operational Fingerprints

This subsection introduces the last study on the number of fingerprints: averaging multiple fingerprints collected at the operational stage. Averaging operational fingerprints is not common since it requires the node to remain static in a position or to have multiple synchronized interfaces to collect data [46]. However, we consider that its study is also relevant to determine the sources of positioning errors.

Table 9 shows the results of (1) traditional deterministic fingerprinting without averaging (rows 1 and 2); (2) fingerprinting where averaging has only been applied to operational fingerprints (rows 3 and 4); (3) fingerprinting where averaging has only been applied to the reference dataset (rows 5 and 7); and (4) fingerprinting where averaging has been applied to the reference dataset and also in the operational stage (rows 6, 8, 9 and 10).

# Ref. FP	Ref. avg.	Op. avg.	Mean Error	Max. Error	MEE	% Cases above the MEE
1	no avg	no avg	3.455 ± 1.969	12.096 ± 1.140	0.707	96.284
10 10 10	no avg no avg no avg	no avg 5 10	$\begin{array}{c} 3.216 \pm 1.837 \\ 2.454 \pm 1.422 \\ 2.344 \pm 1.376 \end{array}$	$\begin{array}{c} 11.692 \pm 1.221 \\ 8.797 \pm 0.860 \\ 8.509 \pm 0.877 \end{array}$	0.707 0.707 0.707	95.754 92.606 91.922
10 10	5 5	no avg 5	$\begin{array}{c} 2.774 \pm 1.605 \\ 1.687 \pm 0.960 \end{array}$	$\begin{array}{c} 10.446 \pm 1.110 \\ 6.152 \pm 0.722 \end{array}$	0.707 0.707	94.051 85.724
10 10	10 10	no avg 10	$\begin{array}{c} 2.753 \pm 1.598 \\ 1.336 \pm 0.762 \end{array}$	$\begin{array}{c} 10.520 \pm 1.141 \\ 4.865 \pm 0.548 \end{array}$	0.707 0.707	94.043 78.268
100 100	5 10	5 10	$\begin{array}{c} 1.534 \pm 0.877 \\ 1.168 \pm 0.672 \end{array}$	$\begin{array}{c} 5.649 \pm 0.602 \\ 4.338 \pm 0.429 \end{array}$	0.707 0.707	83.034 72.644

Table 9. Analysis of averaging training and operational fingerprints in the noisy world using 1-NN.

Table 9 shows that the accuracy is reduced from 3.4 m to 3.2 m by using 10 fingerprints per reference point instead of just 1 (lines 1 and 2); this reductions is about 24 cm. If averaging is applied to the training set only, the accuracy is reduced to \approx 2.76 (lines 5 and 7). If averaging is applied to the operational fingerprints only, the accuracy is reduced to \approx 2.4 m (lines 3 and 4). This finding in the simulation is relevant: it seems that efforts should concentrate on removing the noise present in the operational fingerprints rather than in the training ones according to the results shown. This makes sense since averaging the training or reference fingerprints reduces the number of candidates for the 1-NN algorithm, i.e., the probability of getting the appropriate closest match is high, if the number of candidates (reference fingerprints) is high.

The best results are obtained when averaging is applied to the training and operational fingerprints (lines 6 and 8 in Table 9), with an impressive mean positioning error of 1.687 m (averaging in blocks of 5) and 1.336 (averaging in blocks of 10). The higher the number of averaged fingerprints, the lower the error. However, it seems that there is a lower bound in accuracy of about 1.5 m (when averaging in blocks of 5) and 1.1 m (when averaging in blocks of 10) as shown in the two last cases (lines 9 and 10) for a radio map composed of 100 individual reference fingerprints that have been averaged in blocks of 5 and 10 non-overlapping fingerprints.

Finally, averaging the reference and operational fingerprints not only reduces the mean positioning error, it also reduces large errors and the percentage of cases where the error was higher than the MEE as shown in Table 9 and Figure 22.





Figure 22. Graphical results (mean and maximum positioning errors over 100 runs) for analysing the impact of averaging training and operational fingerprints in the realistic noisy world. (**a**,**b**) 10 fingerprints per reference point without averaging; (**c**,**d**) 10 fingerprints per reference point with averaging in blocks of 5; (**e**,**f**) 10 fingerprints per reference point with averaging in blocks of 10.

4.5.4. General Discussion about the Number of Fingerprints

First, it seems that having a reasonable number of fingerprints per reference point (e.g., ≈ 10 in a realistic scenario with $\sigma = 2$) is enough. Adding hundreds and hundreds of fingerprints per reference point makes no sense in a short period of time: the accuracy is not highly improved since the operational fingerprint is just 1; and the computational costs (at training and operational stages) increase. The accuracy has a lower bound which corresponds to the noise of the operational fingerprint, i.e., if it is an outlier fingerprint, the probability of correctly estimating its position is too low.

Averaging the reference fingerprints improves the accuracy of the indoor positioning systems according to the simulation results. It seems that the noise present in the radio map is significantly reduced and the reference fingerprints are more robust. However, a lower bound in the accuracy persists. It seems that averaging hundreds and hundreds of fingerprints per reference point also makes no sense.

The final step regarding fingerprint density is to increase the number of operational fingerprints and average them. This is not common in indoor navigation solutions since the minimum time between two fingerprints is usually between 1 and 4 seconds according to prior experiments. However, it is feasible if the mobile node remains static in one place or has multiple synchronized interfaces to collect simultaneous fingerprints. Averaging a few operational fingerprints reduces the noise at the operational stage and, therefore, the positioning results are much better since the probability of having an outlier operational fingerprint is much lower.

4.6. Estimation Algorithm

This subsection deals with the study of different estimation algorithms used in fingerprinting.

4.6.1. 1NN vs. kNN

The deterministic kNN algorithm is the base algorithm used by many indoor positioning systems. As has already been mentioned, the position is estimated using the k most similar fingerprints from the reference dataset (or radio map). In the optimistic world, where there is only one fingerprint per reference point, the possible estimations for any fingerprint inside a cell are shown in Figure 23. From the distribution of candidates, the best accuracy is expected to be obtained for k = 3 since the greatest distance from any fingerprint to the closest candidate is the lowest for the four possible values of k. Also, higher values of k should not be suitable since a fingerprint is surrounded by just four fingerprints. A similar behaviour is expected in the quantized world.



Figure 23. Expected solutions of kNN in the optimistic world for k = 1 (red); k = 2 (green and black); k = 3 (blue); k = 4 (black).

Several values of k ({1,3,5,7,...,39,41}) have been tested in the optimistic and quantized worlds, and the results are shown in Figure 24. The plot for the optimistic world shows that the optimal value of k is about 3 to 5 nearest neighbours, whereas 1 seems to be the best k value in the quantized world.



Figure 24. Impact of the number of reference fingerprints in the optimistic (**a**); quantized (**b**) and realistic with $\sigma = 2$ (**c**) worlds. Black, green and red lines correspond to mean, min. and max. accuracy, respectively.

For the realistic world, selecting the best value for *k* is slightly different. First, any operational fingerprint is surrounded by 40 fingerprints (10 fingerprints per reference point) instead of just four fingerprints as in the optimistic and quantized worlds. Second, outlier reference and operational

fingerprints might appear in the scenario due to the nature of the injected noise. For the realistic world, and assuming a noise with $\sigma = 2$, more *k* values have been tested ({1,3,5,7,...,99,101}), and the results are shown in Figure 24. Although the plot shows that the optimal values of *k* are in the range of 7 to 41, k = 7 is good enough and the difference with the other *k* values is not so high.

The previous figures have shown the mean accuracy of the k-NN algorithm for different values of k. Table 10 expands on these figures and shows more information about the maximum positioning error and the percentage of cases above the MEE for the realistic world. According to the table, the lowest mean positioning error is obtained with large k-values. In fact, the best value is provided with k = 41, where the mean positioning error is 2.328 m. Also, the lowest maximum error is provided by large k-values. However, the use of low k-values (e.g., k = 41) is common in the literature.

k-Value	Mean Error	Max. Error	Max. Expected Error	% Cases above the MEE
1	3.218 ± 0.184	11.764 ± 0.111	0.707	95.587
3	2.619 ± 0.146	9.297 ± 0.101	0.707	94.088
5	2.498 ± 0.138	8.753 ± 0.092	0.707	93.485
7	2.442 ± 0.134	8.401 ± 0.079	0.707	93.329
9	2.401 ± 0.132	8.420 ± 0.078	0.707	93.049
11	2.383 ± 0.131	8.394 ± 0.099	0.707	93.120
13	2.377 ± 0.130	8.427 ± 0.101	0.707	93.048
15	2.364 ± 0.129	8.438 ± 0.091	0.707	92.934
17	2.350 ± 0.128	8.101 ± 0.097	0.707	92.928
19	2.352 ± 0.128	8.096 ± 0.084	0.707	92.990
21	2.348 ± 0.128	8.114 ± 0.091	0.707	93.032
23	2.348 ± 0.128	8.064 ± 0.083	0.707	92.917
25	2.342 ± 0.127	8.267 ± 0.102	0.707	92.921
27	2.336 ± 0.127	8.141 ± 0.086	0.707	93.016
29	2.337 ± 0.126	8.069 ± 0.087	0.707	92.994
31	2.335 ± 0.126	7.990 ± 0.072	0.707	93.049
33	2.340 ± 0.126	7.956 ± 0.079	0.707	93.044
35	2.332 ± 0.126	8.150 ± 0.099	0.707	93.068
37	2.337 ± 0.126	8.059 ± 0.111	0.707	92.938
39	2.334 ± 0.126	8.036 ± 0.084	0.707	92.975
41	2.328 ± 0.126	7.942 ± 0.085	0.707	92.948
43	2.337 ± 0.126	7.936 ± 0.083	0.707	92.952
45	2.334 ± 0.126	8.173 ± 0.100	0.707	92.897
47	2.330 ± 0.125	8.124 ± 0.093	0.707	92.924
49	2.333 ± 0.125	7.940 ± 0.085	0.707	93.080
51	2.333 ± 0.125	7.904 ± 0.076	0.707	93.067
53	2.338 ± 0.125	7.981 ± 0.088	0.707	93.024
55	2.332 ± 0.125	8.057 ± 0.091	0.707	93.083
57	2.337 ± 0.125	8.043 ± 0.110	0.707	93.021
59	2.333 ± 0.125	7.924 ± 0.085	0.707	92.955
61	2.330 ± 0.125	7.903 ± 0.088	0.707	93.005
63	2.332 ± 0.125	7.948 ± 0.101	0.707	93.026
65	2.336 ± 0.125	7.992 ± 0.090	0.707	93.082
67	2.336 ± 0.125	8.086 ± 0.090	0.707	93.083
69	2.333 ± 0.125	7.941 ± 0.082	0.707	92.994
71	2.335 ± 0.125	7.961 ± 0.088	0.707	93.026
73	2.339 ± 0.125	7.900 ± 0.071	0.707	93.047
75	2.339 ± 0.125	7.918 ± 0.092	0.707	93.153

Table 10. Analysis of the k-Value for Wi-Fi fingerprinting in the noisy world using k-NN: Mean positioning error; Maximum positioning Error; Maximum expected error (MEE) and percentage of cases above the MEE.

4.6.2. Distance Metrics

k-NN is a well-known algorithm that has been used as the basis in many indoor positioning systems. One crucial parameter of k-NN is the distance or similarity measure, which is used to compare two fingerprints.

In order to analyse the effect of the distance metrics on fingerprinting, we compare the accuracy of 41 alternatives [50] in the optimistic and quantized world first. Table 11 shows the mean accuracy,

the mean maximum error, the percentage of cases above the MEE for the 1 m grid, and the Pearson correlation coefficient between the feature space (FS) distance and the positioning error.

Table 11. Analysis of similarity/distance metrics for Wi-Fi fingerprinting in the optimistic and quantized worlds using 1-NN.

		Optimist	ic			Quantize	d	
Metric	MeanPE	MaxPE	%	Corr.	MeanPE	MaxPE	%	Corr.
additivesymmetric	1.424 ± 0.054	2.526	92.373	0.502	0.759 ± 0.045	2.422	79.517	0.545
bhattacharyya	7.410 ± 0.456	17.003	99.090	0.953	7.914 ± 0.480	18.493	98.637	0.942
camberra	0.407 ± 0.018	1.453	11.681	0.748	0.580 ± 0.032	2.009	55.760	0.356
chebyshev	0.412 ± 0.019	1.166	15.922	0.748	0.499 ± 0.028	1.712	70.725	0.099
cityblock	0.389 ± 0.015	0.943	6.913	0.872	0.493 ± 0.027	1.783	60.349	0.066
cityblockintersect	14.136 ± 0.647	26.610	99.887	NaN	14.133 ± 0.647	26.590	99.894	NaN
clarck	0.632 ± 0.221	40.854	29.657	0.561	0.609 ± 0.036	2.529	57.973	0.441
cosine	0.395 ± 0.016	1.143	9.236	0.701	0.572 ± 0.033	3.020	54.094	0.301
divergence	0.415 ± 0.020	1.658	12.735	0.210	0.598 ± 0.034	2.255	57.303	0.425
euclidean	0.392 ± 0.016	1.029	8.536	0.854	0.492 ± 0.027	1.783	60.362	0.073
fidelity	7.410 ± 0.456	17.003	99.090	0.959	7.914 ± 0.480	18.493	98.637	0.947
harmonicmean	4.980 ± 0.282	11.044	97.962	0.885	5.232 ± 0.289	13.668	97.638	0.831
hellinger	0.388 ± 0.015	1.017	5.999	0.913	0.578 ± 0.032	2.096	55.643	0.366
innerproduct	5.396 ± 0.271	13.862	98.959	0.426	5.850 ± 0.308	14.879	99.006	0.316
jeffreys	0.388 ± 0.015	1.020	6.005	0.874	0.578 ± 0.032	2.096	55.646	0.346
jensen-difference	0.388 ± 0.015	1.017	5.998	0.875	0.578 ± 0.032	2.096	55.643	0.348
jensen-shannon	3.631 ± 0.190	7.988	96.219	-0.873	3.780 ± 0.197	9.551	97.181	-0.764
k-divergence	13.722 ± 0.631	25.780	99.892	-0.229	13.212 ± 0.660	25.780	99.802	-0.431
kullback-leibler	12.064 ± 0.627	24.570	99.655	-0.035	12.356 ± 0.665	24.873	99.456	-0.083
kumar-hassebrook	0.392 ± 0.016	1.019	8.567	0.817	0.568 ± 0.031	1.968	53.938	0.338
KumarJohnson	0.388 ± 0.015	1.019	5.998	0.875	0.578 ± 0.032	2.096	55.662	0.350
lorentzian	0.389 ± 0.015	0.934	6.794	0.873	0.494 ± 0.027	1.788	60.367	0.072
mahalanobis	0.391 ± 0.015	0.974	8.308	0.853	0.554 ± 0.030	1.937	54.514	0.287
matusita	0.388 ± 0.015	1.017	5.999	0.913	0.578 ± 0.032	2.096	55.643	0.366
MaxSymetricChi2	0.389 ± 0.015	1.030	6.083	0.863	0.579 ± 0.032	2.097	55.756	0.350
minkowsky3	0.397 ± 0.016	1.093	10.420	0.827	0.492 ± 0.027	1.783	60.362	0.075
minkowsky4	0.401 ± 0.017	1.111	11.771	0.809	0.492 ± 0.027	1.783	60.362	0.076
minkowsky5	0.404 ± 0.018	1.124	12.717	0.796	0.492 ± 0.027	1.783	60.362	0.077
MinSymetricChi2	0.388 ± 0.015	0.999	5.920	0.877	0.574 ± 0.032	2.033	55.544	0.317
motyka	0.389 ± 0.015	0.940	6.931	0.891	0.525 ± 0.029	1.796	57.392	0.217
neyman	0.389 ± 0.015	1.021	6.143	0.875	0.578 ± 0.032	2.033	55.224	0.331
pearson	0.388 ± 0.015	1.015	5.926	0.874	0.573 ± 0.031	2.037	55.478	0.325
sorensen	0.389 ± 0.015	0.940	6.931	0.891	0.525 ± 0.029	1.796	57.392	0.217
squared	0.388 ± 0.015	1.015	5.993	0.874	0.578 ± 0.032	2.096	55.641	0.348
squaredchord	0.388 ± 0.015	1.017	5.999	0.874	0.578 ± 0.032	2.096	55.643	0.348
Taneia	0.388 ± 0.015	1.021	6.009	0.874	0.578 ± 0.032	2.096	55.616	0.346
topsoe	0.388 ± 0.015	1.017	5.998	0.875	0.578 ± 0.032	2.096	55.643	0.348
VicisSvmetricChi2A	0.415 ± 0.020	1.664	12.811	0.133	0.598 ± 0.034	2.255	57.321	0.414
VicisSymetricChi2B	0.389 ± 0.015	1.026	6.058	0.864	0.579 ± 0.032	2.096	55.721	0.348
VicisSymetricChi2C	0.388 ± 0.015	1.003	5.948	0.877	0.575 ± 0.032	2.031	55.535	0.323
VicisWaveHedges	0.408 ± 0.019	1.492	11.753	0.656	0.581 ± 0.032	2.009	55.875	0.360
wavehedges	0.407 ± 0.018	1.430	11.607	0.757	0.576 ± 0.032	1.921	55.637	0.329

For the optimistic world, it can be observed that only a few distance metrics are not suitable for fingerprinting, since they report a huge average error. Moreover, the *city block* (or *Manhattan*) distance seem to be better alternatives than the *Euclidean distance*, which is the default distance metric in many fingerprinting approaches. Overall, it seems that alternative distances might be better than the *Manhattan* distance in the optimistic world because they provide lower maximum error, lower percentage of cases where the positioning error was lower than the expected threshold and higher correlation (feature space distance vs. geometric distance). The most significant finding is that none of the metrics analysed is able to assign the closest candidate in the geometric space as the best match for all the operational fingerprints. None of the studied distance metrics, some of them widely used in fingerprinting, is able to fit the meaning of the RSSI values.

For the quantized world, the best approach is the *Euclidean distance* according to the mean positioning error, but the *chebyshev distance* provides lower maximum error. However, the correlation

values (distance FS vs. geometric distance) decreased after quantization. Also, the percentage of cases where the error is above the MEE is slightly higher than for the optimistic world. Although the number of unique reference fingerprints in the quantized scenario is \approx 80%, the percentage of cases where the error is higher than expected is about 55%.

Figure 25 shows the relation between the distance in the feature space (Manhattan) of the closest match and the positioning error for the optimistic and quantized worlds through four scatter plots. Two examples are shown for each world: considering all the results provided in the 100 simulations, and considering two representative cells (red and blue circles in the plots) in the 100 simulations.



Figure 25. Relation between the distance in the feature space (Manhattan)—y axis—of the best match and the positioning error—x axis—for the happy world (**a**,**b**) and the quantized world (**c**,**d**). (**a**,**c**) show all the tuples generated in the 100 simulations in all the scenarios; whereas (**b**) and (**d**) show the tuples generated in the 100 simulations in two representative cells.

In the optimistic world, it can be seen that, as expected, there exists a relation between distance in the feature space (Manhattan distance) and geometric distance (error). Although this relation is not perfect, the distance in the feature space might be considered an estimator of the committed error: a low Manhattan's distance corresponds to low errors whereas high values correspond to high errors.

However, when quantization is present in the RSSI values, this assumption is no longer true, as can be observed in Figure 25c,d, where low positioning errors (about 0 m) have large distance in the FS, and low distances in the FS (about 3 dBm) provide accurate positioning. Similar behaviours have been provided by other distance metrics such as *Euclidean distance* and *Mahalanobis distance*, which are widely used in fingerprinting.

For the realistic world, the study is limited to only those distance metrics used in the literature, whose results are shown in Table 12. Figures 26 and 27 show more details for a selection of them.

Table 12.	Analysis	of similarity,	/distance	metrics	for	Wi-Fi	fingerprinting	in the	realistic	world
using 1-NI	N.									

	Optimistic			
Metric	MeanPE	MaxPE	%	Corr.
additivesymmetric	3.594 ± 0.200	12.534	97.790	0.016
cityblock	3.199 ± 0.184	11.642	98.170	-0.054
cosine	3.588 ± 0.205	12.672	97.829	0.024
euclidean	3.312 ± 0.189	11.676	97.756	-0.016
mahalanobis	3.404 ± 0.193	11.729	97.505	-0.005
matusita	3.469 ± 0.196	12.153	97.641	0.023
minkowsky3	3.344 ± 0.191	11.840	97.700	-0.007
minkowsky4	3.347 ± 0.191	11.850	97.713	-0.004
minkowsky5	3.348 ± 0.191	11.847	97.716	-0.005
neyman	3.462 ± 0.195	12.119	97.621	0.014
sorensen	3.391 ± 0.191	12.083	97.692	0.004



Figure 26. Graphical results (mean and maximum positioning errors over 100 runs) for analysing the distance/similarity metric in the noisy world for 1-NN. (**a**,**b**) Manhattan distance; (**c**,**d**) Euclidean distance; (**e**,**f**) Mahalanobis distance; (**g**,**h**) Matusita distance; (**i**,**j**) Neyman distance; (**k**,**l**) Sorensen distance.



Figure 27. Cumulative distribution of the mean (**a**) and maximum (**b**) positioning error (CDF) for eight different distance/similarity metrics. Dashed vertical lines indicate the average error, whereas dotted vertical lines indicate the highest error.

The results provided by the eleven selected metrics are similar, with a mean positioning error between \approx 3.2 and \approx 3.6 m. However, the most important finding is that the Pearson's correlation factor is approximately 0 in all of them; i.e., there is no correlation between the distance in the feature space and the geometric distance (positioning error) for any of the selected distance/similarity metrics.

This last finding shows that it might be complex to estimate the positioning error from the distance in the feature space. Moreover, it shows that the weights calculated in weighted k-NN based algorithms might be not appropriated, since it seems that there is not a perfect relation between the geometric distance and feature space (RSSI) distance for two compared fingerprints. Figure 28 shows the relation between the distance in the feature space (Manhattan) of the closest match and the positioning error in a realistic world scenario ($\sigma = 2$) in a similar way as done for the optimistic and quantized worlds (all cases and three representative cells). According to the figure, the tuples are scattered and there is no correlation between the distance in the feature space and the geometric distance between an operational fingerprint and the closest reference fingerprint.



Figure 28. Relation between the distance in the feature space (Manhattan) of the best match and the positioning error for the realistic world. (**a**) all the tuples generated in the 100 simulations; (**b**) tuples generated in the 100 simulations in three representative cells.

Although the distance in the feature space (RSSI) does not seem to be an indicator of the positioning error, this study goes one step further by showing the mean distance in the feature

space to the closest match (calculated with the distance or similarity metric on the RSSI vectors) in each cell of the scenario. Figure 29 shows the mean distance in the feature space to the closest match for the selected metrics except *minkowsky4*, which has been omitted due to its similarity to the other minkowsky metrics (including *Manhattan* and *Euclidean* distances).



Figure 29. Mean distance in the feature space to the closest match. (a) Manhattan distance; (b) Euclidean distance; (c) Minkowsky3 distance; (d) Minkowsky5 distance; (e) Mahalanobis distance; (f) Matusita distance; (g) Neyman distance; (h) Sorensen distance; (i) Cosine similarity; (j) Additive Symmetric distance.

According to Figure 29, the 10 selected metrics present three different patterns. Figure 29a–e are similar, and the highest mean distance in the feature space is provided when the operational fingerprints are close to the APs (see Figure 7e–f). Figure 29f–i are similar, and the lowest mean distance in the feature space is provided when the operational fingerprints have a high mean RSSI value (see Figure 7c,d). Although Cosine (Figure 7i presents an inverse pattern, it is a similarity metric

(the higher the value, the better), whereas the other ones are distance metrics (the lower the value, the better). Figure 7j presents a different pattern to that of the others, and it seems that the lowest mean distance is provided at the geometric centre of the scenario. It is worth noting that this last distance metric, Additive Symmetric, provides a few distance outliers with values higher than 100 (removed from the figure to improve visibility).

5. Relationship between Observed Large Errors and RSSI Values

5.1. General Discussion about Sources of Large Errors

This paper has explored the main sources of large errors in deterministic fingerprinting: quantization and noise. Also, the relation between large errors and the assumptions commonly made in some fingerprinting parameters is also explored:

- Quantization of RSSI values has introduced uncertainty to radio signal propagation and a difference of just 1 dBm might correspond to a real difference of ≈0 dBm, 1 dBm or ≈2 dBm. The difference of just 1 dBm means a displacement of a few centimeters or many meters depending on the distance with respect to the antenna. Therefore, the real displacement might be almost 0 (real difference of ≈0 dBm) to twice the expected displacement (real difference of ≈2 dBm).
- The presence of noise has increased uncertainty in radio signal propagation, which makes it more difficult to distinguish relatively close fingerprints.
- Generating the radio map using the grid strategy can be considered a source of error since the underpinning of radio propagation is not considered. Moreover, having a dense grid does not imply low positioning errors since there is a lower bound in the IPS accuracy due to the noise present in RSSI values.
- The AP density and distribution are both important. If the distribution is optimized and the distance between access points is low, the probability of having large errors decreases. However, fingerprinting relies on an already deployed Wi-Fi network in many real cases, which might not be optimized for positioning. Moreover, the presence of virtual APs is usually not exploited. This lack of knowledge about the Wi-Fi network should be considered a source of error, since the AP density (number of APs detected) might not correspond to the expected distribution for proper operation of the IPS (e.g., presence of virtual APs, detection of APs placed in adjacent buildings, among others).
- The fingerprint density, i.e., number of fingerprints per reference point, is also important. On the one hand, relying on just one fingerprint per reference point increases the probability of having large errors since the reference database might contain more fingerprint outliers. On the other hand, having many fingerprints per reference point does not have a significant impact on the accuracy of the IPS (there is a lower bound in the system accuracy), but the computational costs at the operational stage to estimate the position increase. The number of fingerprints have to be carefully balanced to have enough samples to provide good accuracy, keeping the computational costs at the operational level reasonable.
- The reference fingerprints in the realistic world are noisy, therefore there might always be large positioning errors due to outliers at the operational stage. One way to mitigate them is to average reference and operational fingerprints. Averaging fingerprints mitigates the injected noise and the probability of obtaining large errors decreases. Although it can easily be done when generating the radio map (reference fingerprints), averaging at the operational stage is not always possible (it requires the mobile node to be static or having multiple synchronized interfaces).
- The deterministic algorithm used for fingerprinting is not perfect and it should also be considered a source of error. The premise "the most similar fingerprint in feature space, the closest one in the geometric space" is not true in most cases (the positioning error is higher than the MEE in more than 90% of cases in the realistic world). Moreover, there is no clear relation between the distance in the feature space and the geometric distance due to the injected noise. This is a fact derived from the quantization and noise present in the RSSI values, but it could also be observed in the

optimistic world (no quantization and no noise), where the closest match in the feature space did not correspond to the closest reference fingerprint in the geometric space in \approx 2% of cases. The deterministic k-NN is a general-purpose rule that does not consider the nature of radio signal propagation.

- None of the studied distance metrics encompasses the logarithmic nature of radio signal propagation and the location of APs. Therefore, the RSSI values are treated as unitless values and the physical underpinning is ignored. A perfect correlation between a distance/similarity metric and geometric distance was also not obtained in the optimistic world. As already mentioned, the distance metrics might also be a cause of large errors according to [28] since they are general purpose.
- Through this comprehensive study, we observed a pattern: large positioning errors are located in the scenario periphery and far from the APs. In other words, fingerprinting provides good positioning accuracy in those areas that are close to an AP and large errors are not expected. Hopefully, the distance to the closest AP can be calculated from the strongest RSSI value in the fingerprint. This is due to the nature of theoretical radio signal propagation: signal degradation is almost depreciable at long distances.

In the real world, fingerprinting is also affected by external features of the environment and issues in the radio signal propagation. Although they are not within the scope of this paper, we enumerate a few of them:

- 1. **Intermittent detection of APs:** Distant APs are sometimes detected at a given position; this intermittent effect affects fingerprinting. To partially avoid it, some authors apply thresholds to remove weak signals in the fingerprints.
- 2. **Device diversity:** This paper has performed a comprehensive study of the sources of large errors with one assumption: the hardware used to collect the fingerprints remains the same for training and operational stages. In the real world, a device with a different antenna gain might be used for positioning. This can also be considered a source of error since the operational fingerprints have a shift in the RSSI values which negatively affects the accuracy [51].
- 3. Environmental dynamics: We have considered a static scenario in our simulations in order to explore the origins of large errors related to the fingerprinting technique. However, one source of large errors is the dynamics of the environment. Positioning accuracy might be highly affected by humidity, differences in the density of people, open/closed doors or in training and operational stages, among other scenario factors [44,52].
- 4. **Changes in the Wi-Fi network:** Many real Wi-Fi fingerprinting solutions rely on an already deployed Wi-Fi infrastructure for communications. Network configuration is susceptible to changes which might have a catastrophic effect if they are not communicated to the IPS administrators, e.g., replacement of APs, reconfiguration of APs or AP rearranging.

5.2. RSSI Values as Indicator of Large Errors

As mentioned in the previous sections, it seems that the simulations showed a pattern in the location of large positioning errors. Most of them were far from the APs, where the maximum RSSI value and mean RSSI values are both low.

In order to conduct an in-depth analysis, the relation between the positioning error and the RSSI statistics (mean and maximum values) was plotted in a scatter plot. Since this kind of plot cannot show the density of tuples with similar values, it was transformed into the heat maps shown in Figure 30. The horizontal axis represents the mean and maximum RSSI value in an operational fingerprint, whereas the vertical axis represents the positioning error by using the 1-NN algorithm. The cell colour indicates how many of the 100,000 analysed fingerprints (1000 operational fingerprints \times 100 simulations) fall into this cell.



Figure 30. Relation between positioning error and RSSI statistics (mean RSSI (**a**) and maximum RSSI value (**b**)) in the simulated scenario considering the 100 repetitions in the simulation.

According to the figures, it seems that the maximum RSSI better represents the upper bound in error positioning in the simulated scenario. On the one hand, the lower the maximum RSSI value, the higher the positioning error that can be obtained. On the other hand, the higher the maximum RSSI value, the more the probability of having a large error decreases. Furthermore, the largest error is lower than 5 m when the max. RSSI is -43 dBm.

Although the simulations mimic realistic scenarios, some additional sources of error that affect the deterministic positioning estimator are not considered (e.g., dynamics in the scenario and other unexpected or "unknown" issues with radio signal propagation). Therefore, to encompass real deployments, an analysis in real world scenarios is shown in Figures 31 and 32. The former figure shows some examples of fingerprinting in real scenarios done by the authors of this paper at their respective institutions. This setup contains one department corridor, one research laboratory [53] and a university library [54] (some details about the datasets and experimental setup can be found in the respective references). The latter figure shows some examples from two external datasets collected by an independent research group [55–58] at the Technical University of Tampere.



Figure 31. Relation between positioning error and RSSI statistics (mean RSSI and maximum RSSI value) in real scenarios. (**a**,**b**) Departamento de Sistemas de Informação, University of Minho; (**c**,**d**) GEOTEC laboratory, Universitat Jaume I [53] (**e**,**f**) Library, Universitat Jaume I [54].



Figure 32. Relation between positioning results in real scenarios and RSSI statistics (mean RSSI and maximum RSSI value). (**a**,**b**) Technical University of Tampere—Building 1 [55,56]; (**c**,**d**) Technical University of Tampere—Tietolato Building [57,58].

It is worth noting that the network infrastructure is an uncontrolled parameter in all the selected real scenarios since it was already deployed, and it is also an "unknown" parameter since information about its configuration and the exact position of some APs are both unknown. In fact, some APs might be placed in the nearby buildings.

The analysis of the relation between positioning error and RSSI statistics is shown in Figure 31 for our scenarios. First, tests were done using a dataset collected at the Departamento de Sistemas de Informação (University of Minho, PT). Although the number of operational samples is much lower compared to the simulation, the relation between RSSI statistics and positioning error is similar to that observed in the simulation (see Figure 30). Large errors occur when the mean and/or maximum RSSI values are weak. Then, a dataset collected from a research laboratory was used. Although the shapes of both relations are similar to the simulation, the positioning errors are much lower since the size of the environment was small and the density of APs in the scenario was very high (i.e., very large errors did not occur); the results obtained at the university library showed the same behaviour. In general, large errors are not expected when the max. RSSI value is high.

The relation between positioning error and RSSI statistics for the TUT scenarios is shown in Figure 32. The relation between RSSI statistics and positioning error is similar to that observed in the simulation (the stronger the max. and/or mean RSSI, the better the positioning accuracy). However, in these two datasets, the trend is less clear and the positioning error is much larger than in the simulations. It is worth mentioning that only one fingerprint per reference point was collected in both

datasets. Moreover, the second dataset was created by means of crowd-sourcing, hence the presence of very large errors, some of them higher than 50 m. Both databases were collected at multi-storey buildings. In general, large errors are not expected when the mean and max. RSSI value is high.

6. Conclusions

A comprehensive analysis of the sources of positioning errors has been introduced in this paper. The quantization of RSSI values and the noise present due to "unknown" factors (uncontrolled factors that cannot be easily modelled such as the impact of humidity or the signal diffraction, abortion, reflection and refraction) are the main sources of errors in deterministic fingerprinting. Quantization adds uncertainty to the relation between the RSSI values and the distance to the AP. This uncertainty is low for those areas close to APs but it becomes larger as the distance to the AP increases. The noise present in the environment, due to unknown signal propagation factors, significantly increases the uncertainty. Thus, the use of deterministic fingerprinting is providing a reasonable mean accuracy for many applications where exact precision is not required. However, based on our experience in previous deployments, there are always a few cases where the errors are extremely large.

These large errors motivated us to perform a comprehensive study in which many controlled (depend on the system developer) and uncontrolled (do not depend on the system developer, such as where the APs are placed) parameters of deterministic fingerprinting were analysed: grid size, density and distribution of APs; density of reference and operational fingerprints and k-NN parameters. The results show that many of our assumptions are not true, according to simulated data:

- Having an extremely dense grid does not guarantee extremely good accuracy.
- Adding more APs beyond a threshold has little positive impact on the positioning accuracy, if the APs are uniformly distributed.
- Having hundreds of fingerprints per reference point does not guarantee extremely good accuracy, if data is not appropriately preprocessed.
- The k-NN basic premise when comparing any two fingerprints, "the lowest distance (or highest similarity) in the feature space, the lowest distance in the geometric space" does not hold true in all cases.
- It seems that there is a lower bound in the accuracy by using deterministic fingerprinting according to our simulations. The most important step now is not to provide a better mean positioning error, but to detect and reduce large positioning errors.

The parameters for deterministic fingerprinting are commonly selected according to the average positioning error and do not consider the underpinning nature of signal propagation. To have a better overview of the positioning errors, this study introduced detailed results that showed the accuracy on each grid cell. The graphical results showed that the accuracy of deterministic Wi-Fi fingerprinting positioning depends on the area where the operational fingerprint is located; this paper has depicted a clear pattern: the areas near to the APs report lower positioning errors according to the simulations.

Finally, we checked this in real deployments (no simulation) and the same pattern was detected in all of them. It is worth noting that additional sources of error were introduced in the real scenarios by using arbitrary reference points (no grid), multiple heterogeneous devices which might have different antenna gains, the presence of mobile hotspots, and the presence of reference fingerprints on multiple floors, among many others. Although the selected real scenarios are more complex than the simulated scenario, it can be stated that the probability of having a large error decreases as the maximum value and mean RSSI value in a fingerprint increase, i.e., a fingerprint with a high maximum and mean RSSI value is less prone to providing a large positioning error.

Acknowledgments: This work has been supported by COMPETE: POCI-01-0145-FEDER-007043 and FCT-Fundação para a Ciência e Tecnologia within the scope of project UID/CEC/00319/2013, by the Portugal Incentive System for Research and Technological Development in the scope of the projects in co-promotion no 002814/2015 (iFACTORY 2015-2018), and by the José Castillejo mobility grant (CAS16/00072).

Author Contributions: Joaquín Torres-Sospedra prepared and run the simulations. Adriano Moreira planned the experiments and provided the parameters for base simulation. Both authors equally contributed in the elaboration of this work.

Conflicts of Interest: The authors declare no conflict of interest. The founding sponsors had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, and in the decision to publish the results.

Abbreviations

The following abbreviations are used in this manuscript:

- AP Access Point
- CDF Cumulative Distribution Function
- FS Feature Space
- FSPL Free Space Path Loss
- GNSS Global Navigation Satellite System
- IPS Indoor Positioning System
- kNN k Nearest Neighbors
- LBS Location Based Services
- MEE Maximum Expected Error
- IPIN Indoor Positioning and Indoor Navigation
- RFID Radio-Frequency Identification
- RSSI Received Signal Strength Indicator
- RTLS Real Time Location System

References

- 1. Filo, K.; Lock, D.; Karg, A. Sport and social media research: A review. Sport Manag. Rev. 2015, 18, 166–181.
- Potortì, F.; Crivello, A.; Girolami, M.; Traficante, E.; Barsocchi, P. Wi-Fi Probes as Digital Crumbs for Crowd Localisation. In Proceedings of the 2016 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Alcala de Henares, Spain, 4–7 October 2016; pp. 1–8.
- 3. Luo, C.; Cheng, L.; Chan, M.C.; Gu, Y.; Li, J.; Ming, Z. Pallas: Self-Bootstrapping Fine-Grained Passive Indoor Localization Using WiFi Monitors. *IEEE Trans. Mob. Comput.* **2017**, *16*, 466–481.
- 4. Yaeli, A.; Bak, P.; Feigenblat, G.; Nadler, S.; Roitman, H.; Saadoun, G.; Ship, H.J.; Cohen, D.; Fuchs, O.; Ofek-Koifman, S.; et al. Understanding customer behavior using indoor location analysis and visualization. *IBM J. Res. Dev.* **2014**, *58*, 3:1–3:12.
- González, J.; Blanco, J.; Galindo, C.; de Galisteo, A.O.; Fernández-Madrigal, J.; Moreno, F.; Martínez, J. Mobile robot localization based on Ultra-Wide-Band ranging: A particle filter approach. *Robot. Auton. Syst.* 2009, 57, 496–507.
- Amini, A.; Vaghefi, R.M.; de la Garza, J.M.; Buehrer, R.M. Improving GPS-Based Vehicle Positioning for Intelligent Transportation Systems. In Proceedings of the 2014 IEEE Intelligent Vehicles Symposium, Dearborn, MI, USA, 8–11 June 2014; pp. 1023–1029.
- 7. Nazemzadeh, P.; Moro, F.; Fontanelli, D.; Macii, D.; Palopoli, L. Indoor Positioning of a Robotic Walking Assistant for Large Public Environments. *IEEE Trans. Instrum. Meas.* **2015**, *64*, 2965–2976.
- Sithole, G.; Zlatanova, S. Position, Location, Place and Area: An Indoor Perspective. In *ISPRS Annals of Photogrammetry, Remote Sensing and Spatial Information Sciences*; Copernicus GmbH: Göttingen, Germany, 2016; Volume III-4, pp. 89–96.
- 9. Jenkins, P.L.; Phillips, T.J.; Mulberg, E.J.; Hui, S.P. Activity patterns of Californians: Use of and proximity to indoor pollutant sources. *Atmos. Environ. Part A Gen. Top.* **1992**, *26*, 2141–2148.
- 10. Haritaoglu, I.; Harwood, D.; Davis, L.S. W⁴: Real-time surveillance of people and their activities. *IEEE Trans. Pattern Anal. Mach. Intell.* **2000**, *22*, 809–830.
- 11. Klepeis, N.E.; Nelson, W.C.; Ott, W.R.; Robinson, J.P.; Tsang, A.M.; Switzer, P.; Behar, J.V.; Hern, S.C.; Engelmann, W.H. The National Human Activity Pattern Survey (NHAPS): A resource for assessing exposure to environmental pollutants. *J. Expo. Anal. Environ. Epidemiol.* **2001**, *11*, 231–252.
- 12. Brasche, S.; Bischof, W. Daily time spent indoors in German homes: A Baseline data for the assessment of indoor exposure of German occupants. *Int. J. Hyg. Environ. Health* **2005**, *208*, 247–253.

- Kjærgaard, M.B.; Blunck, H.; Godsk, T.; Toftkjær, T.; Christensen, D.L.; Grønbæk, K. Indoor Positioning Using GPS Revisited. In Proceedings of the Pervasive Computing: 8th International Conference on Pervasive Computing, Helsinki, Finland, 17–20 May 2010; Floréen, P., Krüger, A., Spasojevic, M., Eds.; Springer: Berlin/Heidelberg, Gremany, 2010; pp. 38–56.
- 14. Liu, H.; Darabi, H.; Banerjee, P.; Liu, J. Survey of Wireless Indoor Positioning Techniques and Systems. *IEEE Trans. Syst. Man Cybern. Part C (Appl. Rev.)* **2007**, *37*, 1067–1080.
- 15. Gu, Y.; Lo, A.; Niemegeers, I. A survey of indoor positioning systems for wireless personal networks. *IEEE Commun. Surv. Tutor.* **2009**, *11*, 13–32.
- 16. Basiri, A.; Lohan, E.S.; Moore, T.; Winstanley, A.; Peltola, P.; Hill, C.; Amirian, P.; e Silva, P.F. Indoor location based services challenges, requirements and usability of current solutions. *Comput. Sci. Rev.* **2017**, *24*, 1–12.
- 17. Bahl, V.; Padmanabhan, V. *Enhancements to the RADAR User Location and Tracking System*; Technical Report; Microsoft Research and Microsoft Corporation: Redmond, WA, USA, 2000.
- 18. Mok, E.; Retscher, G. Location determination using WiFi fingerprinting versus WiFi trilateration. *J. Locat. Based Serv.* **2007**, *1*, 145–159, doi:10.1080/17489720701781905.
- Marques, N.; Meneses, F.; Moreira, A. Combining similarity functions and majority rules for multi-building, multi-floor, WiFi positioning. In Proceedings of the 2012 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Sydney, Australia, 13–15 November 2012; pp. 1–9.
- 20. Mautz, R. Indoor Positioning Technologies. Habilitation Thesis, Department of Civil, Environmental and Geomatic Engineering, Institute of Geodesy and Photogrammetry, ETH Zurich, Zürich, Switzerland, February 2012.
- Torres-Sospedra, J.; Jiménez, A.R.; Knauth, S.; Moreira, A.; Beer, Y.; Fetzer, T.; Ta, V.C.; Montoliu, R.; Seco, F.; Mendoza-Silva, G.M.; et al. The Smartphone-Based Offline Indoor Location Competition at IPIN 2016: Analysis and Future Work. *Sensors* 2017, 17, 577.
- Torres-Sospedra, J.; Moreira, A.J.C.; Knauth, S.; Berkvens, R.; Montoliu, R.; Belmonte, O.; Trilles, S.; Nicolau, M.J.; Meneses, F.; Costa, A.; et al. A realistic evaluation of indoor positioning systems based on Wi-Fi fingerprinting: The 2015 EvAAL-ETRI competition. *JAISE* 2017, *9*, 263–279.
- Held, D.; Levinson, J.; Thrun, S. Precision Tracking with Sparse 3D and Dense Color 2D Data. In Proceedings of the 2013 IEEE International Conference on Robotics and Automation, Karlsruhe, Germany, 6–10 May 2013; pp. 1138–1145.
- 24. Lu, Q.; Liao, X.; Xu, S.; Zhu, W. A Hybrid Indoor Positioning Algorithm Based on WiFi Fingerprinting and Pedestrian Dead Reckoning. In Proceedings of the 2016 IEEE 27th Annual International Symposium on Personal, Indoor, and Mobile Radio Communications (PIMRC), Valencia, Spain, 4–8 September 2016; pp. 1–6.
- 25. Sun, Y.; Xu, Y.; Li, C.; Ma, L. Kalman/Map Filtering-Aided Fast Normalized Cross Correlation-Based Wi-Fi Fingerprinting Location Sensing. *Sensors* **2013**, *13*, 15513–15531.
- 26. Baniukevic, A.; Sabonis, D.; Jensen, C.S.; Lu, H. Improving Wi-Fi Based Indoor Positioning Using Bluetooth Add-Ons. In Proceedings of the 2011 IEEE 12th International Conference on Mobile Data Management, Lulea, Sweden, 6–9 June 2011; Volume 1, pp. 246–255.
- 27. Han, D.; Jung, S.; Lee, S. A sensor fusion method for Wi-Fi-based indoor positioning. *ICT Express* **2016**, 2,71–74.
- Dempster, A.G.; Li, B.; Quader, I. Errors in Deterministic Wireless Fingerprinting Systems for localisation. In Proceedings of the 2008 3rd International Symposium on Wireless Pervasive Computing, Santorini, Greece, 7–9 May 2008; pp. 111–115.
- Lemelson, H.; Kjærgaard, M.B.; Hansen, R.; King, T. Error Estimation for Indoor 802.11 Location Fingerprinting. In Proceedings of the 4th International Symposium Location and Context Awareness (LoCA), Tokyo, Japan, 7–8 May 2009; Choudhury, T., Quigley, A., Strang, T., Suginuma, K., Eds.; Springer: Berlin/Heidelberg, Germany, 2009; pp. 138–155.
- Farshad, A.; Li, J.; Marina, M.K.; Garcia, F.J. A Microscopic Look at WiFi Fingerprinting for Indoor Mobile Phone Localization in Diverse Environments. In Proceedings of the International Conference on Indoor Positioning and Indoor Navigation, Montbeliard-Belfort, France, 28–31 October 2013; pp. 1–10.
- 31. Deng, Z.A.; Wang, G.; Qin, D.; Na, Z.; Cui, Y.; Chen, J. Continuous Indoor Positioning Fusing WiFi, Smartphone Sensors and Landmarks. *Sensors* **2016**, *16*, 1427.
- 32. Berkvens, R.; Peremans, H.; Weyn, M. Conditional Entropy and Location Error in Indoor Localization Using Probabilistic Wi-Fi Fingerprinting. *Sensors* **2016**, *16*, 1636.

- 33. Montgomery, D.C. Design and Analysis of Experiments, 8th ed.; Wiley: Hoboken NJ, USA, 2016.
- 34. Elnahrawy, E.; Li, X.; Martin, R.P. The Limits of Localization Using Signal Strength: A Comparative Study. In Proceedings of the 2004 First Annual IEEE Communications Society Conference on Sensor and Ad Hoc Communications and Networks, Santa Clara, CA, USA, 4–7 October 2004; pp. 406–414.
- Kaemarungsi, K.; Krishnamurthy, P. Modeling of Indoor Positioning Systems Based on Location Fingerprinting. In Proceedings of the IEEE INFOCOM, Hong Kong, China, 7–11 March 2004; Volume 2, pp. 1012–1022.
- 36. Kaemarungsi, K.; Krishnamurthy, P. Analysis of WLAN's received signal strength indication for indoor location fingerprinting. *Pervasive Mob. Comput.* **2012**, *8*, 292–316.
- Youssef, M.; Agrawala, A. On the Optimality of WLAN Location Determination Systems. In Proceedings of the Communication Networks and Distributed Systems Modelling and Simulation Conference, San Diego, CA, USA, 18–21 January 2004.
- Krishnakumar, A.S.; Krishnan, P. On the Accuracy of Signal Strength-Based Estimation Techniques. In Proceedings of the IEEE 24th Annual Joint Conference of the IEEE Computer and Communications Societies, Miami, FL, USA, 13–17 March 2005; Volume 1, pp. 642–650.
- 39. Wallbaum, M. A priori error estimates for wireless local area network positioning systems. *Pervasive Mob. Comput.* **2007**, *3*, 560–580.
- 40. Damosso, E.; Correia, L.M.; European Commission. DGX III "Telecommunications, Information Market and Exploitation of Research". In *COST Action 231: Digital Mobile Radio towards Future Generation Systems: Final Report;* EUR (Series); European Commission: Brussels, Belgium, 1999.
- Haeberlen, A.; Flannery, E.; Ladd, A.M.; Rudys, A.; Wallach, D.S.; Kavraki, L.E. Practical Robust Localization over Large-Scale 802.11 Wireless Networks. In Proceedings of the 10th Annual International Conference on Mobile Computing and Networking (MobiCom '04), Philadelphia, PA, USA, 26 Septemder–1 October 2004; ACM: New York, NY, USA, 2004; pp. 70–84.
- 42. Bahl, P.; Padmanabhan, V.N. RADAR: An In-Building RF-Based User Location and Tracking System. In Proceedings of the IEEE INFOCOM 2000, The Conference on Computer Communications, Nineteenth Annual Joint Conference of the IEEE Computer and Communications Societies, Reaching the Promised Land of Communications, Tel Aviv, Israel, 26–30 March 2000; pp. 775–784.
- Youssef, M.; Agrawala, A.K. The Horus WLAN Location Determination System. In Proceedings of the 3rd International Conference on Mobile Systems, Applications, and Services, Washington, DC, USA, 6–8 June 2005; pp. 205–218.
- 44. Garcia-Villalonga, S.; Perez-Navarro, A. Influence of Human Absorption of Wi-Fi Signal in Indoor Positioning with Wi-Fi Fingerprinting. In Proceedings of the 2015 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Banff, AB, Canada, 13–16 October 2015; pp. 1–10.
- 45. Khalajmehrabadi, A.; Gatsis, N.; Akopian, D. Modern WLAN Fingerprinting Indoor Positioning Methods and Deployment Challenges. *IEEE Commun. Surv. Tutor.* **2017**, *19*, 1974–2002.
- Moreira, A.; Silva, I.; Meneses, F.; Nicolau, M.; Ao, C.P.; Torres-Sospedra, J. Multiple simultaneous Wi-Fi measurements in fingerprinting indoor positioning. In Proceedings of the 2017 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Sapporo, Japan, 18–21 September 2017.
- 47. Correa, A.; Barcelo, M.; Morell, A.; Vicario, J.L. Enhanced Inertial-Aided Indoor Tracking System for Wireless Sensor Networks: A Review. *IEEE Sens. J.* **2014**, *14*, 2921–2929.
- 48. Sklar, B. Rayleigh fading channels in mobile digital communication systems. I. Characterization. *IEEE Commun. Mag.* **1997**, *35*, 90–100.
- 49. Luo, J.; Zhan, X. Characterization of Smart Phone Received Signal Strength Indication for WLAN Indoor Positioning Accuracy Improvement. *J. Netw.* **2014**, *9*, 739–746.
- 50. Cha, S.H. Comprehensive Survey on Distance/Similarity Measures between Probability Density Functions. *Int. J. Math. Models Methods Appl. Sci.* **2007**, *1*, 300–307.
- Laitinen, E.; Talvitie, J.; Lohan, E.S. On the RSS biases in WLAN-based indoor positioning. In Proceedings of the 2015 IEEE International Conference on Communication Workshop (ICCW), London, UK, 8–12 June 2015; pp. 797–802.

- 52. Chen, Y.C.; Chiang, J.R.; Chu, H.H.; Huang, P.; Tsui, A.W. Sensor-Assisted Wi-Fi Indoor Location System for Adapting to Environmental Dynamics. In Proceedings of the 8th ACM International Symposium on Modeling, Analysis and Simulation of Wireless and Mobile Systems (MSWiM '05), Montreal, QC, Canada, 10–13 October 2005; ACM: New York, NY, USA, 2005; pp. 118–125.
- 53. Torres-Sospedra, J.; Montoliu, R.; Mendoza-Silva, G.M.; Belmonte, O.; Rambla, D.; Huerta, J. Providing Databases for Different Indoor Positioning Technologies: Pros and Cons of Magnetic Field and Wi-Fi Based Positioning. *Mob. Inf. Syst.* **2016**, 2016, 6092618:1–6092618:22.
- 54. Mendoza-Silva, G.M.; Torres-Sospedra, J.; Huerta, J. A More Realistic Error Distance Calculation for Indoor Positioning Systems Accuracy Evaluation. In Proceedings of the 2017 International Conference on Indoor Positioning and Indoor Navigation (IPIN), Sapporo, Japan, 18–21 September 2017.
- Shrestha, S.; Talvitie, J.; Lohan, E.S. Deconvolution-Based Indoor Localization with WLAN Signals and Unknown Access Point Locations. In Proceedings of the 2013 International Conference on Localization and GNSS (ICL-GNSS), Turin, Italy, 25–27 June 2013; pp. 1–6.
- 56. Talvitie, J.; Lohan, E.S.; Renfors, M. The effect of Coverage Gaps and Measurement Inaccuracies in Fingerprinting Based Indoor Localization. In Proceedings of the International Conference on Localization and GNSS 2014 (ICL-GNSS 2014), Helsinki, Finland, 24–26 June 2014; pp. 1–6.
- 57. Lohan, E.; Torres-Sospedra, J.; Richter, P.; Leppäkoski, H.; Huerta, J.; Cramariuc, A. Crowdsourced WiFi-Fingerprinting Database and Benchmark Software for Indoor Positioning. *Zenodo* **2017**, doi:10.5281/zenodo.889798.
- 58. Lohan, E.S.; Torres-Sospedra, J.; Leppäkoski, H.; Richter, P.; Peng, Z.; Huerta, J. Wi-Fi Crowdsourced Fingerprinting Dataset for Indoor Positioning. *Data* **2017**, *2*, 32, doi:10.3390/data2040032.



© 2017 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).